Analyzing investments in the power system using optimization modeling Marissa Moultak



# Analyzing investments in the power system using optimization modeling

by

## Marissa Moultak

in partial fulfillment of the requirements for the degree of

#### Master of Science Complex Systems Engineering and Management

Faculty of Technology, Policy and Management

at the Delft University of Technology.

to be defended publicly on Monday June 28th , 2020 at - AM.

Student number:

#### 4939735

Thesis committee:

First SupervisorDr. ir. R.A. VerzijlberghSection Energy and IndustrySecond SupervisorDr. ir. J.H. KwakkelSection Policy AnalysisAdvisorDr. ir. R.M. StikkelmanSection Energy and IndustryExternal SupervisorJ. Peters MScCompany Eneco

An electronic version of this thesis is available at http://repository.tudelft.nl/.



## Preface

This thesis is the final research project to conclude the two year Master in Complex Systems Engineering and Management at the Delft University of Technology.

I would like to express my gratitude to the Delft University of Technology for providing me a challenging and rewarding educational experience, as much in the classroom as outside of the classroom. Not only have I been exposed to countless new educational topics, but I believe a different way of thinking to address the complex problems our society faces. To be immersed in and exposed to Dutch culture provided a unique perceptive that often required me to step outside of what I deemed as normal and comfortable and reevaluate my base assumptions of the inner-workings of society. I believe this was particularly enhanced due to the nature of the Complex Engineering and Management study. As we globally currently are facing quite tumultuous times between Corona virus, climate change with its effects becoming ever more apparent and devastating, and political and social unrest, I believe that the cross-disciplinary, holistic study of the Faculty of Technology, Policy and Management is of extreme pertinence. I am eager and very much looking forward to taking the experiences and overall education I have received and applying it to my future endeavours, with the goal of helping us improve upon the problems we are currently facing, particularly in the energy and climate sectors.

First and foremost, I would like to thank my family and close friends for providing me with an endless amount of support over the past months and for listening to my constant ramble about energy system modeling. I would like to thank Remco, our weekly meetings and your guidance and input was essential - thank you for listening and helping me work through my idea, this was especially helpful given the circumstances. Thank you to Eneco and Jeroen for exposing me to the inner-workings of a large energy generation company (although for a much briefer time than anticipated), providing me with the general topic of the research, and for guidance. Laurens - thank you for first introducing me to electricity markets in your course, your captivating and interactive teaching method made me fascinated in the topic and provided me the basis I needed to perform this thesis. Finally, thank you to Jan for providing useful key insights during our meetings.

This thesis, given the nature of the research and the unforeseen circumstances of this year, has been a very difficult project. Through all the challenges, my utter fascination in the topic never varied, and I feel grateful to have found a project that has captivated and challenged me - solidifying the area I would like to focus my future work. I hope my fascination in this topic translates into an interesting and enjoyable reading experience.

Marissa Moultak

## **Executive summary**

The European electricity system is rapidly changing. Over the next couple of decades, electricity demand is forecast to increase with the electrification of the heat and mobility sectors, and greenhouse emissions are required to decrease dramatically to meet emission reduction targets. The energy transition requires significant investments in new power generation, storage, and transmission technology. However, the complex, uncertain nature of the power system creates challenges for investors. The challenges continue to increase as the energy system transitions to a low carbon system, transitioning away from conventional sources to intermittent renewable energy sources. This transition changes the dynamics of the energy market and paired with unpredictable government climate change policies and regulation, can lead to highly uncertain investment outcomes for generators. Increasing the penetration of wind and solar power in the energy system changes the electricity market dynamics and therefore the resulting electricity prices. Investments in the power sector are long-term investments that require investors to analyze future profitability and risk to make investment decisions. Therefore, understanding and forecasting future market conditions is an essential component for investors to make investments in the system and for policy makers to understand the regulatory framework and market design changes that are necessary to meet emission reduction goals while maintaining a secure and stable power supply.

Under current market arrangements, decarbonization of the power sector will shift the investment environment. With the increasing penetration of VREs, the wholesale electricity price decreases in the short-term due to marginal pricing. The depression in market prices lead to lower revenue for all generators, the merit-order effect. Therefore, the penetration of VREs may drive down investments in the power sector. In an ideal, long-term, perfect market with sufficient CO2 pricing for a desired emissions reduction target, the revenue for VREs will always be enough to cover the total costs. The short-term effects, imperfections in the market, and policy choices can lead to situations where revenues drop below expenses for generators, causing investors to be unable to recoup their investment costs. Such a situation would lead to an under investment in the necessary generation capacity to maintain the security of supply.

Modeling and research is needed to forecast the profitability of investments given the evolving power system and to better understand how power system configurations affect profitability. The purpose of this research is to gain a better understanding into the economic feasibility behind investments in renewable energy generation and storage technologies as the energy system transitions to a low-carbon system. In particular, the research explores how energy optimization modeling paired with the uncertainty analysis, modeling-to-generate alternatives (MGA) can be used to provide insight into and aid in investment decision making. Therefore, the main research question is:

How can energy optimization modeling paired with the uncertainty technique, Modeling to Generate Alternatives be used to help make investment decisions considering the uncertainties in the future electricity system?

The research takes a novel approach to utilizing energy system optimization models (ESOMs) and develops a methodological framework to use ESOMs from the perspective of an investor, to explore the economic feasibility of investments in the future power system in light of emission reduction targets. The method developed includes a partial-equilibrium, myopic, two-step energy system optimization model. The ESOM is paired with the MGA uncertainty analysis to show how optimization modeling can be used to provide a range of possible future power system designs and their resulting range of associated electricity prices and generator and storage dispatch. These results can then be used to determine the profitability of generation or storage technology over their respective lifetimes. From the perspective of a policy maker, the resulting business cases of the investments can be analyzed to shed light on the feasibility of investments in VREs and storage that are necessary to achieve a power system that meets emission reduction targets in a cost efficient manner. The analysis reveals insights into the inner-working of the real world electricity market and with the implementation of the uncertainty analysis MGA, how the imperfections in the market, that cause deviations from optimally, affect longterm profits for investors. As a proof of concept of the methodological framework, a model is developed and used to performed a case study on the Dutch power system. The model is validated with the conceptual, theoretical understanding of energy optimization modeling and verified against historical power system data to ensure it is used appropriately and to explore the limitations of the model. The investor perception of profitability and risk in generation and storage technology in the power system are essential to the energy transition and the realization of government targets. Therefore, the outcomes of this study and others that analyze the investment environment of the power sector are essential not only for investment but for policy makers.

The study develops a method for utilizing energy system optimization modeling from the perspective of an investor. Performing an uncertainty analysis, such as MGA, is imperative when using the energy system optimization modeling to forecast possible future energy systems. The case study performed utilizing the methodological framework developed in the research finds that there is a wide range of possible future near-optimal energy systems, but that profitability of technologies might lead to policy mechanisms or alternative market arrangements being required to achieve emission reduction targets within the near-optimal solution range.

## Contents

Li	st of	Figure	S	xi
Li	st of	Tables		xiii
1	Intro	oductio	on	1
	1.1	Proble	em background	1
		1.1.1	Energy transition - Climate Agreement targets for the power system	1
		1.1.2	Necessity of cost modeling in the changing electricity system landscape	1
	1.2	Proble	em statement	3
	1.3	Resea	arch questions	4
	1.4	Resea	arch contributions.	4
	1.5	Thesis	soutline	6
2	Bac	kgrour	nd & Theory	7
	2.1	Econo	mic theory & the electricity market	7
		2.1.1	Marginal-cost pricing, merit-order, & merit-order effect	8
		2.1.2	Market value	8
	2.2	Energ	y system modeling	9
		2.2.1	Energy system optimization models	10
		2.2.2	Optimization frameworks	12
		2.2.3	Model features: time frame & modeling technique	13
		2.2.4	Optimization modeling & the zero profit rule	14
	2.3	Uncer	tainty in the electricity system	14
		2.3.1	Deviation of real-world power system from energy optimization models	14
		2.3.2	Uncertainty and energy optimization models	15
		2.3.3	Effect of uncertainty on investments	16
		2.3.4	Necessity of uncertainty for forecasting possible energy system futures	16
		2.3.5	Modeling to generate alternatives	17
3	Мос	deling f	ramework	19
	3.1	Backg	round on the developed modeling framework	19

	3.2	Description of the modeling framework			
	3.3	3 Investment optimization			
		3.3.1	Modeling to generate alternatives (MGA) optimization	24	
		3.3.2	Operations optimization	25	
	3.4	Optimi	zation model constraints	25	
		3.4.1	Generator constraints	25	
		3.4.2	Storage unit constraints	26	
		3.4.3	Transmission constraints.	27	
		3.4.4	Nodal energy balance	27	
		3.4.5	Emission constraint.	27	
		3.4.6	Investment calculation	27	
4	Cas	e Study	/	29	
	4.1	Cases	study details	29	
	4.2	Experi	mental setup	30	
	4.3	Model	data	31	
	4.4	Elastic	demand parameters	32	
5	Мос	lel veri	fication & validation	35	
5	<b>Moc</b> 5.1	<b>lel veri</b> t Model	fication & validation	<b>35</b> 35	
5	<b>Moc</b> 5.1	<b>lel veri</b> f Model 5.1.1	fication & validation verification MGA verification	<b>35</b> 35 35	
5	<b>Moc</b> 5.1 5.2	<b>lel veri</b> f Model 5.1.1 Model	fication & validation verification	<b>35</b> 35 35 36	
5	<b>Moc</b> 5.1 5.2	del verit Model 5.1.1 Model 5.2.1	fication & validation         verification	<b>35</b> 35 35 36 36	
5	<b>Moc</b> 5.1 5.2	del verit Model 5.1.1 Model 5.2.1 5.2.2	Fication & validation         verification         MGA verification         validation         Capacity optimization validation         Insights from capacity optimization validation results	<ul> <li>35</li> <li>35</li> <li>36</li> <li>36</li> <li>37</li> </ul>	
5	<b>Moc</b> 5.1 5.2	del verit Model 5.1.1 Model 5.2.1 5.2.2 5.2.3	Fication & validation         verification         MGA verification         validation         capacity optimization validation         Insights from capacity optimization validation results         Operations optimization & electricity price validation	<ul> <li>35</li> <li>35</li> <li>36</li> <li>36</li> <li>37</li> <li>37</li> </ul>	
5	<b>Moc</b> 5.1 5.2	del verit Model 5.1.1 Model 5.2.1 5.2.2 5.2.3 5.2.4	Fication & validation         verification         MGA verification         validation         capacity optimization validation         Insights from capacity optimization validation results         Operations optimization & electricity price validation         Insights from operations optimization validation	<ul> <li>35</li> <li>35</li> <li>36</li> <li>36</li> <li>37</li> <li>37</li> <li>38</li> </ul>	
5	<b>Moc</b> 5.1 5.2 <b>Cas</b>	del verit Model 5.1.1 Model 5.2.1 5.2.2 5.2.3 5.2.4 e study	Fication & validation         verification         MGA verification         validation         validation         Capacity optimization validation         Insights from capacity optimization validation results         Operations optimization & electricity price validation         Insights from operations optimization validation         r: Results & analysis	<ul> <li>35</li> <li>35</li> <li>36</li> <li>36</li> <li>37</li> <li>37</li> <li>38</li> <li>41</li> </ul>	
5	Moc 5.1 5.2 Cas 6.1	del verit Model 5.1.1 Model 5.2.1 5.2.2 5.2.3 5.2.4 e study Optima	fication & validation         verification         MGA verification         validation         capacity optimization validation         Insights from capacity optimization validation results         Operations optimization & electricity price validation         Insights from operations optimization validation         r: Results & analysis         al results	<ul> <li>35</li> <li>35</li> <li>36</li> <li>36</li> <li>37</li> <li>37</li> <li>38</li> <li>41</li> <li>41</li> </ul>	
5	Moc 5.1 5.2 Cas 6.1 6.2	del verit Model 5.1.1 Model 5.2.1 5.2.2 5.2.3 5.2.4 e study Optima Result	fication & validation         verification         MGA verification         validation         capacity optimization validation         Insights from capacity optimization validation results         Operations optimization & electricity price validation         Insights from operations optimization validation         r: Results & analysis         al results         s for the Netherlands	<ul> <li>35</li> <li>35</li> <li>36</li> <li>36</li> <li>37</li> <li>37</li> <li>38</li> <li>41</li> <li>41</li> <li>45</li> </ul>	
5 6 7	Moc 5.1 5.2 Cas 6.1 6.2 Disc	del verit Model 5.1.1 Model 5.2.1 5.2.2 5.2.3 5.2.4 e study Optima Result	fication & validation         verification         MGA verification         validation         validation         Capacity optimization validation         Insights from capacity optimization validation results         Operations optimization & electricity price validation         Insights from operations optimization validation         r: Results & analysis         al results         s for the Netherlands         n, research limitations & further research	<ul> <li>35</li> <li>35</li> <li>36</li> <li>36</li> <li>37</li> <li>37</li> <li>38</li> <li>41</li> <li>41</li> <li>45</li> <li>53</li> </ul>	
5 6 7 8	Moc 5.1 5.2 Cas 6.1 6.2 Disc Con	del verit Model 5.1.1 Model 5.2.1 5.2.2 5.2.3 5.2.4 e study Optima Result cussion	fication & validation verification MGA verification MGA verification Capacity optimization validation Capacity optimization validation Insights from capacity optimization validation results Operations optimization & electricity price validation Insights from operations optimization validation <b>r: Results &amp; analysis</b> al results s for the Netherlands <b>r. research limitations &amp; further research n &amp; recommendations</b>	<ul> <li>35</li> <li>35</li> <li>36</li> <li>36</li> <li>37</li> <li>37</li> <li>38</li> <li>41</li> <li>41</li> <li>45</li> <li>53</li> <li>57</li> </ul>	
5 6 7 8 9	Moc 5.1 5.2 Cas 6.1 6.2 Disc Con Refi	del verit Model 5.1.1 Model 5.2.1 5.2.2 5.2.3 5.2.4 e study Optima Result cussion clusion	fication & validation verification MGA verification MGA verification validation Capacity optimization validation Insights from capacity optimization validation results Operations optimization & electricity price validation Insights from operations optimization validation r: Results & analysis al results s for the Netherlands , research limitations & further research h & recommendations	<ul> <li>35</li> <li>35</li> <li>36</li> <li>36</li> <li>37</li> <li>37</li> <li>38</li> <li>41</li> <li>41</li> <li>45</li> <li>53</li> <li>57</li> <li>61</li> </ul>	

B Research flow chart	67
C Validation dispatch curves	69
Bibliography	71

## List of Figures

Historical and forecasted (2020-2030) yearly electricity production per source and total greenhouse gas emissions in the Netherlands (CBS, 2019).	2
Merit-order effect. The solid black line denotes the supply curve without wind energy and the dashed line is the supply curve with the addition of wind energy to the merit-order. Source: (Bahar and Sauvage, 2013)	8
Classification of energy system models. Adapted from: (Prina et al., 2020)	10
Relationship between investment and operation optimization models. Source: (Pérez-Arriaga, 2013)	14
Modeling framework for determining profitability of generation and storage technologies using energy system optimization modeling and the MGA uncertainty technique	20
Schematic of each full scenario. This is performed for the investment optimization case (optimal case) and the various MGA cases.	21
Supply and demand curves. The dashed triangle is the utility.	23
Example of how the minimum and maximum value for a decision variable are found using MGA. Decision space is multi-dimensional so this is repeated for each decision variable.	24
Graphical representation of the set-up for the optimal run and each MGA run.	30
Bar chart of installed capacities by generator and storage types for both the Netherlands and Germany for 2015.	32
Price duration curve for the 2015 historical prices and the 2015 model prices	33
Bar chart of installed capacities by generator and storage types for both the Netherlands and Germany of the historical, optimal, and each MGA alternative.	37
Bar chart of installed capacities by generator and storage types for both the Netherlands and Germany of the 2010 and 2019 historical values and the 2019 optimal values and values for each MGA run from the model.	38
Bar chart comparing the installed total generator, storage, and transmission capacities for both the Netherlands and Germany for the optimal solution in 2030, 2040, and 2050. The generator and storage capacities are shown by the bars (left axis) and the transmission capacities are given with the points on the graph (right axis).	41
	Historical and torecasted (2020-2030) yearly electricity production per source and total greenhouse gas emissions in the Netherlands (CBS, 2019).         Merit-order effect. The solid black line denotes the supply curve without wind energy and the dashed line is the supply curve with the addition of wind energy to the merit-order. Source: (Bahar and Sauvage, 2013)         Classification of energy system models. Adapted from: (Prina et al., 2020)         Relationship between investment and operation optimization models. Source: (Pérez-Arriaga, 2013)         Modeling framework for determining profitability of generation and storage technologies using energy system optimization modeling and the MGA uncertainty technique.         Schematic of each full scenario. This is performed for the investment optimization case (optimal case) and the various MGA cases.         Supply and demand curves. The dashed triangle is the utility.         Example of how the minimum and maximum value for a decision variable are found using MGA. Decision space is multi-dimensional so this is repeated for each decision variable.         Graphical representation of the set-up for the optimal run and each MGA run.         Bar chart of installed capacities by generator and storage types for both the Netherlands and Germany for 2015.         Price duration curve for the 2015 historical prices and the 2015 model prices.         Bar chart of installed capacities by generator and storage types for both the Netherlands and Germany of the historical, optimal, and each MGA alternative.         Bar chart of installed capacities by generator and storage types for both the Netherlands and Germany of the 2010 and 201

6.2	Bar chart comparing the total generation for the Netherlands and Germany per technol- ogy for the optimal solution in 2030, 2040, 2050 of a average operations year (year used in the investment optimization).	42
6.3	Bar chart comparing the total generation for both the Netherlands and Germany per technology for the optimal solution over the entire modeled time horizon: 2030-2059.	43
6.4	Bar chart comparing the capacities for both the Netherlands and Germany for each al- ternative generated using MGA for 2030, 2040, and 2050.	43
6.5	Bar chart comparing the capacities for both the Netherlands and Germany for each al- ternative generated using MGA for 2030, 2040, and 2050. The grey plotted line markers denote the total maximum potential renewable energy capacities in the Netherlands and Germany.	44
6.6	Bar chart comparing the total capacities the Netherlands for the optimal solution in 2030, 2040, and 2050.	45
6.7	Bar chart comparing the built out capacities in the Netherlands for each alternative generated using MGA for 2030, 2040, and 2050.	46
6.8	Bar chart comparing the built out capacities in the Netherlands for each alternative gen- erated using MGA for 2030, 2040, and 2050. The grey plotted line markers denote the total maximum potential renewable energy capacities in the Netherlands.	47
6.9	The top two graphs are stacked dispatch curves for all generation for a week in August from the 2030 model results. The bottom two graphs are the corresponding LMP curves over the same week in August. The graphs on the left are for the Netherlands and on the right are for Germany.	47
6.10	Histogram of all the LMP for all runs for each investment time period. Electricity prices greater than 200 EUR/MWh are all in the last bin	48
6.11	Price duration curves for 2030, 2040, and 2050 of the optimal scenario.	49
6.12	Histogram of NPV for technologies built in 2030 with bootstrapping 1,000 possible com- bination of NPVs within each 10-year investment period	50
6.13	Boxplot for each run of NPV for technologies built in 2030 with bootstrapping 1,000 pos- sible combination of NPVs within each 10-year investment period.	51
B.1		67
C.1	Price duration curves for each 10-years (one year for each operations optimization run with a different demand and weather year) of alternative MGA runs (blue curves) plotted against the historical price duration curve for 2019 (green curve).	69

## List of Tables

2.1	Optimization frameworks adapted from (Hirth, 2013)	13
A.1	Greenhouse gas emissions from electricity generation for the Netherlands and Germany for 2015, 2020, 2030, 2040, 2050 (United Nations, 2020; Federal Ministry for the Environment and Safety, 2020; CBS, 2019; Umweltbundesamt, 2020).	63
A.2	Potential renewable energy capacities for Germany and the Netherlands used in model (Brown, Hörsch, and Schlachtberger, 2018)	63
A.3	Techno-economic data used in model	64
A.4	Fuel price, carbon intensity, and value of loss load forecasts for 2030, 2040, and 2050 in Europe.	65
A.5	Historical installed generator capacities in the Netherlands and Germany for 2010, 2015, and 2020.	65

## Nomenclature

#### Abbreviations

Energy system optimization modeling
Market value
Value of Loss Load
Variable Renewable Energy Sources
Levelized Cost of Electricity

- VC Variable Cost
- FC Fixed Cost
- MGA Modeling to Generate Alternatives
- LMP Locational marginal price

#### **Mathematical terms**

#### Sets & indices

Set	Index	Description	
G	g	Generators	
S	S	Storage units	
L	I	Transmission lines	
Т	t	Dispatch time step	
Y	у	Year	
I	i	Investment period	
Ν	n	Nodes	

#### **Decision variables**

Primal variables					
Symbol	Description	Units			
K <sub>n,g</sub>	Power capacity of generator g at node n	MW			
$k_{n,g,t}$	Dispatch of generator g at node n at time t	MWh			
$H_{n,s}$	Power capacity of storage unit s at node n	MW			
$h_{n,s,t}^+$	Charge/storing of storage unit s at node n at time t	MWh			
$h_{n,s,t}^{-}$	Discharge/dispatch of storage unit s at node n at time t	MWh			
soc <sub>n,s,t</sub>	State of charge of storage unit s at node n at time t	MWh			
$F_l$	Power capacity of transmission at line I	MW			
$f_{l,t}$	Flow of line I at node n at time t	MWh			
$d_{n,b,t}$	Demand of consumer b at node n at time t	MWh			

Dual variables

Symbol	Description	Units
$\lambda_{(n,t)}$	marginal price at node n at time t	Eur/MWh
$\mu_{CO_2}$	CO <sub>2</sub> price	Eur/ton CO <sub>2</sub>

#### Parameters

Symbol	Description	Unit
$c_{n,g/s}$	Annualized capital cost of generator g or storage unit s at node n	Eur/MW/year
Cl	Annualized capital cost of transmission line I	Eur/MW/year
<i>o<sub>n,g</sub></i>	Operating cost of generator g at node n	Eur/MWh
$\eta_{n,g}$	Efficiency of generator g at node n	$MW_e l/MW_t h$
$d_{n,t}$	Electric load at node n for time t	MWh
w <sub>t</sub>	weight of each time step t	hours/year
$\alpha_{l,n,t}$	Direction of power flow of transmission line I at node n for time t	
$CAP_{CO_2}$	CO2eq.emissions cap	tons $CO_2$ eq.
$e_{n,g}$	Emissions for generator g at node n	ton CO2eq./MWh
$r_{n,s}$	Time a storage unit can discharge at full output capacity	hours
$\eta^+_{n,s}$	Charging efficiency	%
$\eta_{n,s}^-$	Discharging efficiency	%
$U_{n,b,t}$	Utility of consumer b at node n at time t	EUR/h
$K_{n,g}^{min}$	Minimum installable capacity of generator g at node n	MW
$K_{n,g}^{max}$	Maximum installable capacity of generator g at node n	MW
$H_{n,s}^{min}$	Minimum installable capacity of storage unit s at node n	MW
$H_{n,s}^{max}$	Maximum installable capacity of storage unit s at node n	MW
$F_l^{min}$	Minimum installable capacity of transmission at line I	MW
$F_l^{max}$	Maximum installable capacity of transmission at line I	MW

## Introduction

#### 1.1. Problem background

#### 1.1.1. Energy transition - Climate Agreement targets for the power system

European governments recently provided their 'National Energy and Climate Plan' for the period 2021-2030. The plans provide detailed outlines on how emission reduction targets will be met in each respective country to help achieve the global climate goals outlined in the 2015 Paris Agreement of limiting global warming to less than two degrees Celsius above pre-industrial levels. To achieve the long term decarbonization targets, the necessary changes to the entire energy system are substantial during the 2020-2050 period (Capros et al., 2018). The Dutch Climate Agreement details the plan to achieve the government's greenhouse gas emission reduction target of reducing greenhouse gases by 49% compared to 1990 levels (Netherlands Ministry of Economic Affairs (EZK), 2019). To transition to a sustainable energy system that has a high level of security and economic competitiveness, the share of the electricity generated from renewable energy sources must significantly grow (Child et al., 2019; Gioutsos et al., 2018). The target for the electricity sector is to reduce carbon emissions by at least 20.2 Mt by 2030, and in order to achieve these carbon reductions, the government plans for the electricity production from renewable energy sources to be at least 84 TWh in 2030 (Netherlands Ministry of Economic Affairs (EZK), 2019). According to the 2019 Netherlands Climate and Energy Outlook, the historical and forecasted values for total yearly greenhouse gas emissions from the electricity sector and the yearly amount of electricity produced from renewable sources are shown in Figure 1.1 below. In 2050, the government aims to achieve a 95% reduction in greenhouse gas emissions relative to the 1990 levels and for the electricity production to be carbon neutral (CBS, 2019). The amount of renewable energy sources will have to continue to increase well past the 2030 installed renewable targets to achieve the 2050 goals.

#### 1.1.2. Necessity of cost modeling in the changing electricity system landscape

Due to the variable, distributed nature of VRE, integrating high shares of VRE into the grid makes the network more complex and presents unprecedented challenges in planning, regulating, and operating



Figure 1.1: Historical and forecasted (2020-2030) yearly electricity production per source and total greenhouse gas emissions in the Netherlands (CBS, 2019).

the electricity system in a safe, stable, balanced, and financially feasible manner (Hörsch et al., 2018; Johnston et al., 2019; Kraan, Kramer, and Nikolic, 2018; Orgaz, Bello, and Reneses, 2018). Alongside the transition to VRE, the electricity system has been unbundled and liberalized (Hörsch et al., 2018). The liberalized, restructured power system introduces competition in the market and requires a balance between regulation and competition to provide cost efficiency and achieve emission reduction targets while maintaining security of supply (Botterud, 2003). The introduction of competition introduces decentralized decision making, competitive prices, profit maximization, partial information, and high levels of uncertainty compared to centralized decision-making, regulated tariffs, cost minimization, complete information, and limited levels of uncertainty with a heavily regulated system (Botterud, 2003). As the World Energy Council stated, it is "a time of unprecedented uncertainty for the energy sector" (World Energy Council, 2014). Due to the essential importance of electricity to our entire society, it is of the up-most importance that the entire electricity generation and delivery process is done reliably and cost-efficiently (Morales et al., 2014).

The changing energy system landscape increases the uncertainties in the system and therefore makes the associated investment decisions more difficult. Investments in the power system are subject to a number of factors, including technology advancements, government policy, the power market, and the national economy (Agency, 2007). The uncertain power market dynamics and unpredictable government climate change policy, make the investment cash flows increasingly more uncertain (Agency, 2007). The uncertainty, long time frame, and high cost regarding electricity generation and transmission assets make the associated investment decisions very challenging (Conejo et al., 2016). The high capital cost but low operating costs of renewable energy has and will continue to reduce the role of the market in guiding investments (Newbery et al., 2018). The integration of the highly variable renewable energies "will be challenging without substantial modification to the current '1st generation' market design" (Newbery et al., 2018). Liberalized electricity markets are designed based on merit order, where dispatchable electricity generation are ranked based on marginal cost. The market assumptions in such a market are undermined by large scale deployment of renewables, as renewables have zero marginal cost and cannot be dispatched (Kraan, Kramer, and Nikolic, 2018). With increasing penetration of low-carbon technologies, energy market revenues from the wholesale market price based on short-term marginal cost pricing under current market arrangements are not enough to stimulate the necessary amount of investments in new low-carbon generation capacity in a timely, low cost manner (Directorate-General for Economic and Financial Affairs, 2015; IEA, 2016).

To avoid the problems facing security of supply due to insufficient funding, 'the missing money prob-

lem', the market has to be redesigned to to support the high shares of renewables that are necessary to meet government targets (Newbery et al., 2018). Governments have to intervene by setting subsidy regimes and use capacity mechanisms to overcome market imperfections and internalize negative externalities (IEA, 2016; Newbery et al., 2018; Shahnazari et al., 2015; Winkler et al., 2016). The new market design needs to provide better price signals, incentives for investments, and greater system flexibility (Newbery et al., 2018). In the long run the goal is to create market arrangements that result in a sufficient amount of revenue for the necessary investments to take place without government intervention (Directorate-General for Economic and Financial Affairs, 2015). Regulatory regimes, market design, and system operation must be developed simultaneously with demand response, storage, and low carbon technology deployment to ensure electricity security in a cost-efficient manner while enabling the overall low-carbon transition (IEA, 2016).

Rigorous analysis, sophisticated models, and flexible simulations are required to help inform government policy and private company investments regarding new electricity generation, storage, and transmission in the increasingly complex future energy system (Conejo et al., 2016; Hilpert et al., 2018). Energy analytics are essential to make informed energy decisions regarding energy system design, implementation, and operation (Bazilian et al., 2012; Deng and Lv, 2020). Modeling the uncertainty in the energy system is crucial for the investment decision-making process for new power generation by generation companies (Conejo et al., 2016). In addition, dynamic and stochastic modeling can help governmental decision-making on a regulatory level by providing insight into the performance of the power system under different policies, regulations, and market designs (Botterud, 2003). Participants in the electricity system need to adapt to the changing regulatory environment and need to appropriately account for the increase in risk exposure. Planning methods need to be adjusted to account for the changes in the system and mathematical models need to be further developed to assist the decision making process in the new system landscape (Botterud, 2003). To help overcome these challenges, the research primarily focuses on determining the profitability of investment business cases in the range of near optimal energy system design solutions that achieve emission reduction targets. Investors can use this information to help in the decision making process in the evolving power system. In addition, the insights into how different power system configurations affect the favorability of investment business cases can provide useful guidance to policymakers regarding necessary policies to ensure the required investments needed to reach climate goals.

#### 1.2. Problem statement

A significant amount of research has been dedicated to the advancement and development of energy optimization models. A number of sophisticated open source models have been developed and have been deemed mature enough in comparison to commercial/ proprietary tools to be used for serious use (Groissböck, 2019). Typically, energy system optimization models are used to determine a single cost-optimal energy system (the optimal installed generation, storage, and transmission capacities) given a set of constraints (e.g. CO2 emissions limits). The uncertainty pertaining to the future and the inability of mathematical models to accurately represent the complexity of the energy system cause optimal solutions to have limited significance, lack robustness and can even mislead decision makers by providing false precision in the future energy system optimization models has been identified to be lacking and one of the major challenges of energy system optimization modeling. (Yue et al., 2018)

identifies that uncertainties inherent in the model structure and input parameters in energy system optimization modeling are "at best underplayed and at worst ignored". Sub-optimal solutions may be favorable for reasons outside of purely cost, including public acceptance, land-use conflicts, ease of implementation (Neumann and Brown, 2019). The real-world energy system transition has been shown to not follow the cost-optimal solution but rather, fall within the range of near-optimal energy systems (Trutnevyte, 2016). To be able to account for this, an uncertainty technique, modeling-to-generate alternatives (MGA), has recently been applied to energy system optimization models (Neumann and Brown, 2019; DeCarolis et al., 2016; Price and Keppo, 2017). MGA explores the decision space to generate the maximally different near-optimal solutions within a defined cost slack from the optimal solution. Therefore, MGA provides a range of near optimal power system configurations. As the energy system has been shown to fall within the range of near optimal solutions, this research explores how energy system optimization modeling paired with MGA can be used by the investor. The investor in the system needs to understand the profitability of different investments in the range of possible future energy system configurations.

The recent advancements and sophistication of open source energy system optimization models paired with the uncertainty of investments sparks the question of how these advanced tools can be used from the perspective of an investor to explore the economic feasibility of investments, given the government emission reduction targets and whether the power systems solutions give by these ESOMs provide a feasible investment environment. This research looks to help contribute to two main areas currently lacking in energy system optimization modeling - accounting for uncertainty in energy system optimization modeling and identifying the profitability of of technologies in the range of various possible future power systems.

#### 1.3. Research questions

How can energy optimization modeling paired with the uncertainty technique, Modeling to Generate Alternatives, be used to help make investment decisions?

- 1. How does energy optimization modeling work and how are electricity prices determined in energy system modeling?
- 2. How can the real electricity market be represented in energy optimization modeling?
- 3. How can modeling-to-generate-alternatives (MGA) be used to generate a range of electricity price and dispatch scenarios?
- 4. How can the range of near optimal solutions generated by modeling to generate alternatives provide economic profitability insights for investors in the power system?
- 5. Using the method developed in a case study, what insights can be provided for the economic feasibility of investments in the Dutch electricity system?

#### 1.4. Research contributions

1. Develop a methodological framework for how energy system optimization paired with MGA can be used from the perspective of the investor, to generate insights into the profitability of investments.

- 2. Determine the profitability of investments in the range near optimal energy system solutions.
- 3. Contribute to the exploration of structural uncertainty in energy system optimization modeling.
- 4. Explore how the range of short-term realistic electricity system operation conditions can be accounted for in a long-term model.

The research develops a novel method for analyzing the economic feasibility of investments considering the long-term future power system. To the best of our knowledge, this research is the first to utilize energy system optimization modeling paired with the uncertainty analysis, MGA to examine the profitability of investments in the power system. The research builds upon existing energy system optimization modeling by uncovering a framework for how the rich and developed existing ESOMs can be used from the perspective of an investor in the system and how economically feasible investments are the in the range of near-optimal solutions that can be generated by ESOMs.

Typically, ESOMs are used to determine the cost optimal energy system or more preferably, using uncertainty analyzes, can determine the range of near-optimal energy systems, given a set of constraints. However, the cost optimal or near optimal solutions determined by energy system modeling does not necessarily ensure the economic feasibility of investments, as costs of technologies can outweigh benefits in the least-cost system (Ifzal, 1991). This can be problematic in determining the feasibility of achieving these future energy systems. In liberalized power systems, economic incentives must exist to achieve the adequate investment in generation and storage technologies. This is of particular importance as electricity demand increases due to the electrification of the heat and mobility sectors and as power systems are decarbonized to achieve climate targets. These two factors translate into a need of a significant amount of investments in the power system over the coming decades. Therefore, the method developed in this research provides insights not only for investors but insights for policy makers who need to understand the economic feasibility of the various near-optimal power systems is essential for policy makers to understand the necessary policy to help ensure the realization of an energy system within these near-optimal alternatives.

In addition, the research builds upon past research of utilizing the MGA analysis with energy system optimization modeling to explores the range of near-optimal power system designs. Price et. al identifies that the majority of energy-environment-economy models focus primarily on parametric uncertainty, largely neglecting structural uncertainties of the model (Price and Keppo, 2017). This research helps to contribute to the limited exploration of structural uncertainty in energy system models. In addition, the research examines how to incorporate detailed short-term variability in the power system in a study that examines the long-term, a main challenge that (Ringkjøb, Haugan, and Solbrekke, 2018) highlighted in their review of energy and electricity system modeling tools with high renewable penetrations. The research aims to help overcome this challenge and contribute to the representation of short-term variability in long-term studies by coupling of the investment optimization model with the operations (dispatch) optimization model.

This research purely focuses on the structural uncertainty, developing a modeling framework that uses modeling to generate alternatives to assess the profitability of investments in the future power system. The scope is limited to structural uncertainty due to time and computation constraints and due to the identified particular lack of research into structural uncertainty in energy system modeling Price and Keppo, 2017. However, assessments using energy system optimization models should carefully ac-

count for both the parametric and structural uncertainty (DeCarolis, 2011). Therefore to provide robust insights to decision makers, a parametric uncertainty analysis should be included in future research. The parametric uncertainty analysis can be done using various uncertainty techniques (i.e. Monte Carlo or stochastic optimization). These uncertainty techniques would need to be done to the set of input parameters used in the modeling framework, resulting in the developed framework being done for each iteration of the Monte Carlo or stochastic optimization runs.

#### 1.5. Thesis outline

Chapter 2 provides the necessary background and theory into the electricity system, economics, energy optimization modeling, and uncertainty modeling to provide a basis for the remainder of the report. Chapter 3 introduces and provides a detailed description of the model developed in this research, including the mathematical formulation of the model components. In Chapter 4, the details of the case study that is performed using the developed modeling framework are introduced. Chapter 5 details the model verification and validation to ensure the usability of the model. In Chapter 6 the results and analysis of the case study are explored. In Chapter 7, the research, the research limitations, and further areas of research are discussed. Chapter 8, concludes the main body of the report with main findings and recommendations. Chapter 9 provides a personal reflection of the process of carrying out the thesis. Finally, the data used in the model can be found in the Appendices.

 $\sum$ 

## **Background & Theory**

#### 2.1. Economic theory & the electricity market

Economic theory is needed to understand the foundation and working of the electricity market. In theory, competitive markets are the best structure to obtain economic efficiency. In other words, the law of supply and demand is identified to be the best mechanism to allocate production resources and determine the prices for goods and services (Ventosa, Linares, and Pérez-Arriaga, 2013). To progress towards a more economically efficient system design, since the 1990s, power systems in many countries have been liberalized and electricity markets have been instated to introduce competition in generation and retailing (Ventosa, Linares, and Pérez-Arriaga, 2013). Therefore, to reliably provide power at the lowest societal cost while mitigating market power and enabling transparency, electricity is bought, sold, and traded in a competitive, wholesale market (Milligan et al., 2017). In wholesale electricity markets, marginal-cost pricing is used to set the electricity price. Perfect competitive markets in long-term equilibrium follow the zero profit rule – all producers fully recover their costs through the market price set by marginal cost pricing (Ventosa, Linares, and Pérez-Arriaga, 2013; Milligan et al., 2017; Brown and Reichenberg, 2020). Under perfect markets, the optimal solution can be achieved by investors making investments based on profit (Brown, 2020).

In reality, due to imperfect markets and nonconvex costs, marginal-cost pricing alone does not ensure that generator revenues cover all capital expenses (Brown and Reichenberg, 2020; Milligan et al., 2017). The following six attributes make electricity markets deviate from being perfectly competitive (Milligan et al., 2017):

- 1. externalities (i.e. emissions)
- 2. inelasticity of demand
- 3. market power
- 4. network is a public good
- 5. price caps
- 6. time dependency of electricity (no large scale storage)

The imperfections in the real market affect the economic performance of markets and cause them to

deviate from the economic efficiency expected from ideal markets. These imperfects lead to several market failures in electricity generation (Newbery et al., 2018). Regulatory intervention is required to help account for the imperfections in the real market and are needed to ensure:

- 1. fair competition
- 2. ensure all market players have adequate information
- 3. free entry for all market participates
- 4. all cost and benefits (including externalities) are accounted for

#### 2.1.1. Marginal-cost pricing, merit-order, & merit-order effect

If the power market is structured to follow the merit-order, the marginal cost of the most expensive generator required to operate to satisfy demand at each point in time determines the marginal cost of electricity; this generator is called the "marginal generator".

Energy generation capacity bids into the market at the marginal cost of generation. As the marginal costs of VRE are close to zero, VRE bid into the electricity market at near zero marginal cost, the bottom of the merit order. As shown in 2.1, the introduction of VRE to the market shifts the merit-order curve to the right and can lead to a cheaper marginal generator, therefore decreasing the price the market clears. This is called the merit-order effect.



Figure 2.1: Merit-order effect. The solid black line denotes the supply curve without wind energy and the dashed line is the supply curve with the addition of wind energy to the merit-order. Source: (Bahar and Sauvage, 2013)

#### 2.1.2. Market value

Market value of a particular generator is defined as the average revenue per unit of energy sold or in other words, the portion of the base-load price that the generator is able to capture. Market value differs per generator due to different generator availability and cost. Market value is calculated by the average market price weighted by the hourly production:

$$Market \ value = \frac{\sum_{t} \lambda_{t} K_{g,t}}{\sum_{t} K_{g,t}}$$
(2.1)

 $\lambda_t$  : market price at time t

 $K_{a,t}$ : Dispatch of generator at time t

Market value of VRE is crucial for both policy makers and renewable generators to perform economic assessments of VRE in the electricity system.

(Brown and Reichenberg, 2020) shows how under perfect competitive markets in long-term equilibrium, the market value of each generator equals  $(MV_g)$  the levelized cost of electricity  $(LCOE_g)$  (Brown and Reichenberg, 2020):

$$MV_g = LCOE_g \tag{2.2}$$

LCOE is the sum of all costs (investment, fuel, operation and maintenance costs) averaged over each unit of energy generated.

(Brown and Reichenberg, 2020) finds that if CO<sub>2</sub> price is the policy instrument used to incentive the increase share of wind and solar, VRE revenue will always be sufficient to cover their full costs under perfect competitive markets in long-term equilibrium. In the short-term, the integration of VRE impacts the electricity market price, due to the merit-order effect as described above. Without market interventions (i.e. CO2 prices) the average market price is reduced due to the lower marginal costs of VRE (Brown and Reichenberg, 2020). The lower electricity prices reduce the revenue of all generators but particularly those of wind and solar. Wind and solar depress the market prices at times they are producing and therefore cannibalize their own market values (Brown and Reichenberg, 2020). As a result, the market value of VRE can drop below their LCOE, leading to lower revenues than are necessary to recover capital cost investments. Policy interventions are needed to help overcome these challenges and ensure that investments are made into the required capacity to meet climate targets while maintaining system reliability. The transition to a clean power system require changes to the regulatory framework and the market design (Abrell et al., 2019). Policy makers need to provide a framework to enable industry to achieve climate targets at a reasonable cost while also ensuring security of supply (Newbery et al., 2018). To help determine the necessary policy interventions, sophisticated modeling tools are necessary. One of these modeling techniques, energy system optimization modeling, the modeling technique utilized in this research is described in the next section.

#### 2.2. Energy system modeling

Flexible, in-depth, sophisticated modeling tools are vital to help inform government policy and private company investments regarding new electricity generation plants and transmission lines in the increasingly complex future energy system (Conejo et al., 2016; Hilpert et al., 2018; Pereira, Ferreira, and Vaz, 2016). By providing in-depth analyses into the optimized future structure of the power system, models help to ensure the transition to a low carbon power system, aligned with the climate targets, is achieved reliably and cost effectively (Tash, Ahanchian, and Fahl, 2019). There are a wide range of energy system models.

Figure 2.2, provides an overview of the classification of energy system models and shows the further details of bottom-up models (Prina et al., 2020). From the diagram, it can be seen that energy system models can be divided into three types: top-down models, hybrid models, and bottom-up models. There are three types of bottom-up models: optimization models, simulation models, and accounting models (Fleiter, Worrell, and Eichhammer, 2011). Simulation models and optimization models have typically been used previously for investment analysis. As the focus of this research is to explore the use of energy system optimization modeling for the profitability of power system investments in the

future energy system, the remainder of the section details the types and components of energy system optimization modeling. Energy system optimization models have been used widely to provide critical data driven insights into energy policy at the regional, national, and global scale (Yue et al., 2018).



Figure 2.2: Classification of energy system models. Adapted from: (Prina et al., 2020)

#### 2.2.1. Energy system optimization models

Focusing in on optimization models, they can be classified as either dispatch optimization or investment optimization. Dispatch optimization models are short-term models that optimize the dispatch of a given set of installed capital stock, determining how to best utilize the available technology (i.e. dispatch of the available generators). The decision variables for the short-term, operations optimization are the hourly generation, storage, and transmission. Investment optimization models determine the optimal system investments, and therefore the optimal generation portfolio to supply the demand under the given constraints (Brown, 2020). They can either optimize over the medium-term or the long-term. Medium-term optimization performs a brownfield optimization, it takes the existing infrastructure as a given and allows for endogenous investments and dis-investments. The medium-term optimization considers the existing infrastructure as sunk cost and therefore, the costs of existing infrastructure are disregarded from the optimization. A long-term optimization is a green-field optimization, it determines the optimal system from scratch. The decision variables for the investment (capacity expansion) planning optimization, are the installed capacities and hourly dispatch of generators, storage, and transmission. The investment op-

timization technically has an operations optimization embedded within the optimization, such that the optimization finds the optimal set of investments in generation capacity that, when operated optimally, leads to the minimum total system costs (Pérez-Arriaga, 2013), as is shown in Figure 2.3.

Perfect foresight models perform one optimization over the entire modeled time horizon and therefore assume that investment decisions take into account the entire modeled time horizon and have full information of all system parameters including future costs, prices, and constraints (Poncelet et al., 2016). Perfect foresight assumes that all data about the future is known from the start of the optimization and the optimization is simultaneously performed over the entire model time horizon. For the myopic case, the model is split into several time frames and each time period is optimized separately. Myopic models perform a sequence of optimizations, where decision makers are assumed to have perfect information for each period that is optimized and no information outside of the foresight window. Investors make decisions with limited knowledge of the future. The myopic approach can represent the short-time frames that occur in the real world decision making (Decarolis et al., 2017). In addition, the myopic approach can help to reveal the possible transition technologies.

An optimization model consists of an objective function, decision variables, model parameters, constraints, index sets, and input data sets (Dreier and Howells, 2019). Techno-economic optimization models can be designed to be deterministic, stochastic, or a hybrid between the two. Deterministic models have a fixed set of parameters, stochastic models have random parameter values, and therefore, a hybrid model would have a combination of random and fixed parameter values. The objective of an ESOM is either to maximize social welfare or minimize total system costs, given a set of technological, resource, environmental, and policy constraints (Berntsen and Trutnevyte, 2017). For the case of welfare maximization, the objective is to maximize total economic welfare, which is the sum of the consumers' and generators' surplus. The results of the optimization problem give the optimal set of generator, storage, and transmission capacities that when operated optimally result in the maximum total economic welfare for the energy system modeled. The objective function for a welfare maximization problem is:

$$Maximize \begin{pmatrix} \text{Total economic} \\ \text{welfare} \end{pmatrix} = \sum \begin{pmatrix} \text{Consumer utility} \\ (\text{value function}) \end{pmatrix} - \sum \begin{pmatrix} \text{Generator costs} \\ (\text{supply function}) \end{pmatrix}$$
(2.3)

In cases where the demand is assumed to be inelastic, the optimization objective can be cost minimization. Inelastic demand means that the consumer utility remains unchanged with changes to the electricity price. Therefore, since the consumer utility is constant, it does not affect the overall optimization and the overall optimization problem can be formulated as a cost minimization optimization with the objective to minimize the total annual system costs assuming inelastic demand. For the cost minimization optimization, the total annual system costs are minimized to find the long-term cost-optimal electricity system. To minimize the total annual system cost, the sum of the annualized investment costs (the annualized capital costs for generation, storage, and transmission capacities) and the annual operational costs for the snapshots of time identified in the model are minimized.

$$Minimize \begin{pmatrix} \text{Total annual} \\ \text{system cost} \end{pmatrix} = \sum \begin{pmatrix} \text{Annualized} \\ \text{capital cost} \end{pmatrix} + \sum \begin{pmatrix} \text{Annual} \\ \text{operating cost} \end{pmatrix}$$
(2.4)

The results of the optimization problem give the optimal set of generator, storage, and transmission capacities that when operated optimally result in the lowest overall costs for the annual demand data given to the model.

The objective function of both the welfare maximization and the cost minimization problems are subject to a number of constraints, including generator constraints, storage unit constraints, transmission constraints, emission constraint, and the nodal energy balance. The generator, storage unit, and transmission constraints ensure that the generators, storage units, and transmission lines operate in a technically feasible manner and fall within their minimum and maximum bounds. The emission constraint imposes a limit on the overall emissions in the system and the nodal energy balance ensures that the energy at each node is perfectly balanced at each point in time.

The electricity price results from the nodal energy balance as the marginal electricity price is the dual variable of the nodal energy balance. The dual variable represents the incremental change in the optimal solution value of the objective function as the right hand side of the constraint is relaxed by one unit. Dual variables are also referred to as shadow price and they signify the amount consumers are willing to pay for an additional unit of the given resource (electricity in this case). For utility based objective functions, the dual variable is the marginal utility of relaxing the constraint. For cost based objective functions, the dual variable is the marginal cost of relaxing the constraint. For the case of the cost minimization optimization problem, the optimal solution is the total system cost. For the nodal energy balance in the cost minimization optimization problem, the dual variable represents the amount the total system cost increases to generate one more unit of electricity, the marginal cost of electricity. For an investment cost minimization problem without load shedding, the marginal cost at the maximum demand would be the cost of installing another unit of generation capacity of the marginal generator. This results in a very large dual variable to the nodal energy balance or in other words, a very large marginal cost of electricity at the maximum demand over the modeled time period, much larger than is experienced in the real world electricity market. To be able to overcome this the demand can be modeled as elastic or in the case of inelastic demand, load shedding can be incorporated into the model. For the case of inelastic demand, where the demand does not change in response to changes in the price, the demand time series is a parameter provided to the model. The demand must be perfectly met at each time step throughout the model, except for at very high marginal utility. At very high marginal utility, the utility is the demand multiplied by the value of loss load, the maximum amount the average consumer is willing to pay to avoid loss of power. The load shedding can instead be represented by a dummy generator that has no fixed cost and very high variable costs, set at the value of loss load. In this work, to obtain more realistic electricity prices in the operations optimization, the demand is modeled to be elastic. In this work, the demand is modeled as elastic to more accurately represent the real world.

#### 2.2.2. Optimization frameworks

The optimization problem can be performed under a short, medium, or long-term framework. Each framework refers to a different assumption regarding the capital stock (Hirth, 2017). In the short-term, the existing generation and transmission are given, there are no changes to the infrastructure. The short-term optimizes over only the operation of the system, the investments are not considered in the optimization problem, and therefore is a dispatch model. For the medium term case, the existing infrastructure is provided as a starting point to the model but endogenous investments and disinvestments are possible. Both the medium-term and the long-term require the addition of the investment in capacity. The medium-term optimizes over both the investments in new technology and the operation of the system. Since the existing infrastructure is sunk cost, it is disregarded from the optimization problem. The long-term is a green-field optimization, meaning that all existing infrastructure is disregarded and

the system is built from scratch. It is important to distinguish between the different types of optimization. Table 2.1 describes the differences between the different types of optimization. As (Hirth, 2013) identifies, the market value of VRE is dependent on the modeling assumptions made regarding the existing installed capacities. Since the market value directly influences the profitability of VREs, the modeling assumptions greatly impact the profitability.

	Short-term	Medium-term	Long-term
	(Dispatch)	(Transition Brownfield)	(Greenfield)
Time frame optimized	Short	Medium	Long
Type of optimization	Operation	Hybrid	Investment
Ontimizes	Operations VC	Operations VC &	Operations VC &
Optimizes		New capacity investment FC	All capacity investment FC
Existing capacity	Included	Included	Not included
		VC of existing	
Cost sovings of VPE	VC (existing	convl sources	VC & FC
COST SAVINGS OF VRE	convl sources)	FC (avoid new	(convl sources)
		convl sources)	
Long torm profits	±/	0/-: existing capacity	0
	+ <b>/-</b>	0: new capacity	U

Table 2.1: Optimization frameworks adapted from (Hirth, 2013).

VC = variable cost; FC = fixed cost; convl = conventional<sup>1</sup>

#### 2.2.3. Model features: time frame & modeling technique

In addition to the optimization frameworks specifying how the capital stock is dealt with in the optimization problem, The dispatch optimization and investment optimization align with the different timescales decisions are made in the energy system. These decision timescales are the operational, short-term and investment, long-term decisions. A main difficulty in an infrastructure planning model is the combination of different time scales. The planning of new infrastructure requires strategic decisions that have time horizons of many years, where as to gain insight into the the infrastructure's performance and profitability requires the modeling of the short-term operation of the system (Kaut et al., 2013). Both levels of time scales have associated uncertainty. As can be seen in Figure 2.3, the operational model is nested in the investment model. The sequence of decisions and the time-frame of the various decisions that can be made are very important to understand for power system development (Pérez-Arriaga, 2013).

ESOMs optimize the installed capacity given a representative year within each time period of demand and weather data. Many ESOMs consider a limited amount of time-steps to limit the computational power and time requirements. Limiting the number of time-steps can become problematic when high shares of VREs are considered in the model (Decarolis et al., 2017). An investment ESOM can be paired with a dispatch model to help overcome the limitations.

	↓ F	or a given demand
INVESTMENT MC	DEL:	
Min total cost T Investment K	C = F C (K) +	V C*(K, D)
WHERE:	GIVEN D	GIVEN K*
	OPERATION MODEL V C (K*, D) = min V C (K*, F) Fuel F	
	SUBJECT TO N f (K*, F	MEETING DEMAND: =) = D

D: demand; K: installed capacity; F: fuel supply K\*(Q): optimal investment; VC\*(K,Q): optimal operation

Figure 2.3: Relationship between investment and operation optimization models. Source: (Pérez-Arriaga, 2013)

#### 2.2.4. Optimization modeling & the zero profit rule

As is discussed in Section 2.1, in perfectly competitive markets, the profit for all installed capacity, and therefore all investors, is exactly zero in the long-run. This results from the fact that in a perfectly competitive market if profit is being made new competition will enter the market and sell electricity at a lower price until the profit disappears. If the price is too low and investors are not recovery their costs, investors will leave the market until the price is high enough to recover costs (Pérez-Arriaga, 2013). The deviations of the electricity market in reality lead to deviations from the zero profit rule. A deterministic long-term ESOMs with perfect foresight follows the zero profit rule, with all investors making precisely enough to cover their costs. The long-term profits per optimization type are shown in Table 2.1.

#### 2.3. Uncertainty in the electricity system

#### 2.3.1. Deviation of real-world power system from energy optimization models

Many least-cost, bottom-up models of the electricity sector assume a perfectly competitive power market with perfect foresight (such as PyPSA-Eur, MARKAL) (Loulou, Goldstein, and Noble, 2004). In long-term equilibrium, the assumption and modeling of perfect competition and foresight lead to all system investors making precisely zero profit in the long run (Brown and Reichenberg, 2020). The economic theory perspective shows if investors are making a profit new actors will enter the market, driving profits to zero and if investors are losing money, market players will leave the market until losses no longer occur. Market equilibrium results in zero net profit for all participants, all participants receive a revenue that precisely covers all their costs. Often, demand is modeled as being inelastic until it reaches a threshold where the acquisition of electricity is more than the utility. This non-served energy or loss load is modeled in energy optimization models as another generation technology with very high variable costs and no fixed costs, this value is know as the value of loss load (VoLL). The inclusion of VoLL ensures that in an investment ESOM, that assume a perfectly competitive market with perfect foresight, will result in zero profit for all market investors. The VoLL is the amount the average customer is willing to pay to avoid in interruption in electricity supply (IEA, 2016). It is the loss in consumer surplus as a result of loss of load or in other words the loss of socio-economic activity or costs that results when electricity is not supplied to customers. In practice, power prices rarely reach the theoretical VoLL which can cause the market to provide insufficient revenues for investors leading to

the "missing money" problem (Petitet, Finon, and Janssen, 2017). This is due to a variety of reasons including operational price caps, political unacceptability of high prices, and system operator behavior at peak hours (Petitet, Finon, and Janssen, 2017). In addition, in practice VoLL is difficult to accurately estimate.

If the regulator is not seeking any specific objectives (such as security of supply, reliability, or environmental targets), the market price with sufficient scarcity pricing is sufficient to ensure adequate profits to generators. However this is often not the case. Adequate electricity supply is generally a very politically sensitive topic and therefore, regulators typically like to ensure generous security and reliability margins. Mandating energy capacity above optimal market determined values leads to situations that deviate from the economically efficient power system design (Pérez-Arriaga, 2013). Policies to achieve certain security, reliability, and environmental objectives include VRE installed capacity targets, minimum consumption of a certain fuel, emission caps, minimum installed capacity margin above peak. Achievement of these policy objectives, requires an additional form of payment must be supplied in addition to the market price. Failure of the regulator to adequately remunerate for these security, reliability, or environmental objectives can lead to a missing money problem for investors.

#### 2.3.2. Uncertainty and energy optimization models

Following the classification used by DeCarolis2011, this paper classifies uncertainty into two distinct categories: structural uncertainty and parametric uncertainty. The structural uncertainty is defined as the uncertainty in the structure of the model due to the imperfect mathematical relationships and parametric uncertainty, the uncertainty in the model parameters (Decarolis et al., 2017). Structural uncertainty in the model stem from the fact that optimization models can not fully encompass the complexity of the real world, there will always be additional real world constraints and objectives that are not captured in the model (Decarolis et al., 2017).

Often a single optimal solution is generated for energy system optimization analyzes. As mathematical models never provide a perfect representation for the real world, modelers are frequently unaware or unable to model all practical constraints, and key future input parameters have a significant amount of uncertainty, singular cost optimal solutions provide merely an approximation for the real-world optimal solutions (Voll et al., 2015). The uncertainty pertaining to the future causes optimal solutions to have limited significance and can even mislead decision makers by providing false precision in the future energy systems (DeCarolis et al., 2016; Voll et al., 2015). Trutnevyte2016a performed a retrospective study and found that cost-optimal scenarios deviate significantly from the historical cumulative total system costs (Trutnevyte, 2016). The modeled cost optimal system did not approximate the developments that occurred in the real world (Trutnevyte, 2016). The cost optimal scenarios ignore the large amount of uncertainty arising from deviations from cost optimality. The large amount of parametric and structural uncertainty translates to an extremely low probability of selecting one scenario that will precisely represent the real-world transition (Trutnevyte, 2016). Therefore, a bounded analysis should be used to explore the possible future scenario space and analyze the extremes/ bounds. A large number near-optimal scenarios should be considered to provide an "envelope of predictability" and capture the real-world transition (Trutnevyte, 2016).

Reviews of energy system optimization models find the uncertainty in the structure of energy system optimization models and the input parameter data to be under developed (Foley et al., 2010; Yue et al., 2018). Uncertainty is a vital component of energy system models to ensure that they provide re-

sults and insights that are robust and enable effective, well supported decision making (Yue et al., 2018). ESOMs should be used to explore alternative system configurations and suggest various alternatives to achieve policy goals under deep uncertainty. Decisions with deep uncertainty cannot be based on the most likely, optimal scenarios or forecasting of future developments (Papadelis, Flamos, and Psarras, 2013). Robust strategies, strategies that perform well compared to alternatives across a wide range of possible future scenarios, are key for evaluating alternative decisions under conditions of deep uncertainty (Davis, Bankes, and Egner, 2007; Lempert et al., 2006; Papadelis, Flamos, and Psarras, 2013; Walker, Lempert, and Kwakkel, 2013). Therefore, uncertainty analyses must be effectively communicated to decision makers to aid in robust decision making (Price and Keppo, 2017). The uncertainty inherent to the modeling process must be properly explored and accounted for (Price and Keppo, 2017).

One method for helping to account for the structural uncertainty in energy systems is Modeling to Generate Alternatives (MGA). By exploiting the near-optimal solution space, MGA provides a range of possible near-optimal solutions that helps facilitate robust decision-making, especially when considering other system constraints or variables that are not considered in the model. The MGA analysis provides decision makers with the required features, features that are common across all near-optimal solutions, the features that can be decided upon, features that are different between the near-optimal solutions, and the features that must be avoided, features that are not in present in any of the near-optimal solutions (Voll et al., 2015).

#### 2.3.3. Effect of uncertainty on investments

The uncertainties in the electricity system cause the variability in profitability of investments. Krey2009 identifies that current investment decisions need to take into account the future risks of the uncertainties in the energy sector including basically the entire energy chain, energy demand, and future environmental policies. Investors in the capital-intensive energy system face not only market driven uncertainties but are particularly vulnerable to high degree of policy uncertainties (Fuss et al., 2008).

(Botterud, 2003) identifies that long-term uncertainties and their impact on optimal investment decisions are often underrepresented in long-term decision making. To fill this gap, dynamic models are created to provide a new framework for long-term investment analysis in new power generation capacity and price analysis given the restructured power systems. (Conejo et al., 2016) identities that investments in generation facilities require a long-term view, careful and comprehensive accounting of uncertainties, and large-scale optimization problems. The book goes on to further identify that comprehensive modeling and consideration of the uncertainty in the energy system is critical for investment decision-making. The large amount of uncertainty and risk in the electricity system results from its high sensitivity to sociocultural, geopolitical, economic, environmental, and technical parameters.

#### 2.3.4. Necessity of uncertainty for forecasting possible energy system futures

To use energy optimization modeling to provide insight for investment decision making, the range of future possible outcomes for the design of the electricity system must be considered. Typically, energy system optimization models determine the cost optimal system of a set time period, given a set of constraints.

Policy makers are very unlikely to make policy decisions based purely on cost-optimality, other factors

are considered in driving policy (Price and Keppo, 2017). The energy system has numerous stakeholders who do not have perfect foresight and each have their own objectives and non-cost related preferences(Price and Keppo, 2017). Although it is highly unlikely that the real-world energy transition will occur in precisely a cost optimal manner, it is expected to be driven by cost considerations (Price and Keppo, 2017). Therefore, the real-world energy system could be represented by a near-optimal cost solution.

The use of cost optimization for modeling the energy transition is based on the idea of a central planner, one centralized decision maker, whose goal is to maximize social welfare. Maximizing social welfare is achieved by maximizing both consumer and producer surplus with elastic demands. Equivalently maximizing social welfare can be transformed into minimizing total system cost, which is the negative of total surplus. Under inelastic demand this becomes purely minimizing producer total costs (Trutnevyte, 2016). However, in reality, there is not one centralized decision-maker. The energy system is composed of a variety of actors with various levels of decision power (Trutnevyte, 2016). In addition, cost optimization is used on the basis of partial equilibrium, that energy supply-demand equilibrium is met. Cost optimization models assume a perfect market with rational actors that behave in a cost optimal manner. Although costs are a key driver to the energy transition, the real world energy transition is very likely to not follow a cost-optimal pathway (Trutnevyte, 2016).

The real-world transition is suggested to follow the most investable path rather than the least cost path, since private energy companies are the ones who make the investments, not the government (Trutnevyte, 2016; Gross, Blyth, and Heptonstall, 2010). Investments in installed capacity are determined by expected return, which is based on risk regarding costs and returns (Gross, Blyth, and Heptonstall, 2010). Return risks are highly dependent on the electricity price fluctuations (Gross, Blyth, and Heptonstall, 2010). Policy makers currently mainly focus on the levelized cost of electricity per technology, considering only price risk (Trutnevyte, 2016; Gross, Blyth, and Heptonstall, 2010). However the real-world transition follows the most investable path suggests that policymakers should consider revenue risk to increase the effectiveness of policy making.

All of these reasons provide evidence as to why a range of near optimal solutions need to be used to provide insights into the future energy system. As the real-world transition is expected to deviate from the cost-optimal solution, in order to use energy optimization modeling to help support investment decision making, we must consider a range of near-optimal solutions and parametric uncertainty. This is supported by the ex-post study performed by (Trutnevyte, 2016) of near-optimal energy system modeling compared to the real-world energy transition in the UK from 1990-2014, that found that near-optimal systems can encapsulate the real-world energy transition and therefore should be used to gain general insights of the bounds of possible technology deployment. A method for generating the near-optimal systems is modeling to generate alternative (MGA), described in detail in the next section.

#### 2.3.5. Modeling to generate alternatives

Due to the large uncertainty about the future, ESOMs should be used to identify patterns across many different model runs to produce insights rather than singular projections (Decarolis et al., 2017; Neumann and Brown, 2019). To address structural uncertainty, Modeling to Generate Alternatives (MGA) is a technique for determining near-optimal solutions by systematically exploring the decision space. MGA was first introduced by (Brill, Chang, and Hopkins, 1982) in water and management planning to determine a set of significantly different alternative solutions that are feasible and perform well with

respect to the modeled objective. The ESOM formulation is modified to find a range of possible solutions that are maximally different in decision space but near the optimal solution in solution space (Decarolis et al., 2017). The optimal solution is used as a starting point and is relaxed by a certain cost increase to explore the near-optimal solution space (DeCarolis et al., 2016; Yue et al., 2018). MGA is a complement to methods that address the range of parametric uncertainty (Neumann and Brown, 2019). Near optimal solutions can help to account for unmodeled circumstances (i.e. unforeseen or unmodeled risks) (Yue et al., 2018).

MGA explores the range of technically diverse future electricity system configurations that meet model objectives within a set cost deviation from the cost optimal (least cost) system. The analysis can be used to provide a set of rules that must be satisfied to ensure the total system costs are within a set range of the cost optimal solution. Therefore, it can help guide policy by determining features that are required to keep the system costs within predefined ranges.

The can often be numerous solutions to a mathematical model of a problem that lead to very similar results with respect to the modeled variables but vary significantly in the decision space (Brill, Chang, and Hopkins, 1982). These alternative solutions can be better than other solutions with respect to unmodeled system components (Brill, Chang, and Hopkins, 1982). The methodology behind MGA analysis is further discussed in Chapter 3.
3

# Modeling framework

## 3.1. Background on the developed modeling framework

This research looks to examine the profitability of investments in the actual power market and therefore requires analysis of realistic market conditions. Many ESOMs determine the optimal installed capacity of the power system over a certain time horizon under the assumptions that the market is perfect, competitive and deterministic. Under these assumptions, all of the capacity installed in the optimal system follows the zero profit rule, all investors receive precisely the necessary revenue to cover their expenses. In reality, as described in Section 2.1, there are many attributes of the electricity market that cause it to deviate from being a perfect, competitive market. The modeling framework presented utilizes the energy system optimization modeling technique paired with the MGA uncertainty technique over a long-term time frame, considering both long-term and short-term decision making. The use of the MGA uncertainty technique, the variation in weather and demand years used in the two step optimization process, the myopic optimization, and the elastic modeling of demand result in a deviation from the zero profit rule and aims to provide a more realistic representation of the real world. By providing a more realistic representation of the real world, the model is used to explore the profitability of investment decisions.

# 3.2. Description of the modeling framework

To account for these deviations from a perfect market and to represent the resulting variety of possible future business cases for each generation technologies, a medium-term, investment ESOM paired with the MGA analysis is developed and used to generate a variety of possible power system designs for a set of investment periods (i.e. 2030, 2040, 2050).

The investment medium-term optimizations are only calculated for each investment stage, where installation and decommissioning are assumed to only occur at the start of each investment stage. To be able to simulate a range of possible power system designs, the MGA technique is utilized to explore the decision space by minimizing and maximizing the installed capacities and allowing the system to be up to 10% more expensive than the optimal system. Then, the range of possible power system designs are used as the basis for a short-term, dispatch operations optimization for a variety of weather and demand years to represent the short-term variation that occurs in the power system. A two-step optimization is performed for each investment period to address the two main time frames of decision-making in the energy system, long-term decisions where investments are made and short-term operational decisions where dispatch is determined. The full model flow diagram is shown in Figure 3.1.



Figure 3.1: Modeling framework for determining profitability of generation and storage technologies using energy system optimization modeling and the MGA uncertainty technique.

Figure 3.2 provides a linear schematic for each full alternative run to show the relationship between investment periods and operation periods. The diagram provides a detailed representation of each of the dashed boxes in the modeling framework shown in Figure 3.1. Following the notation in (Kaut et al., 2013), circles denote investment decision stages (investment periods) and squares represent operational decision stages (operation periods). The model consist of three investment periods; investment decisions are assumed to only occur at these stages. The installed capacities are assumed to remain constant over the remainder of the time period. Using the installed capacity determined in the investment stage, a set of operation optimizations are performed given different historical demand and weather data. Each operations optimization is based on the capital stock determined in the respective investment stage, represented in the figure by the fact that investment nodes (grey circles) are the parent nodes of the set of operations optimizations. As can be seen, it is assumed that the operational decisions and their outcomes do not affect future investment or operational decisions. Therefore, the

results from each investment stage feed directly into the consecutive investment stage, without considering the operational outcomes.



Figure 3.2: Schematic of each full scenario. This is performed for the investment optimization case (optimal case) and the various MGA cases.

A main challenge in formulating an optimization model over a long time horizon is the balance between adequately accounting for the complexities of the power system while being able to be solved within computational and time limitations. The developed modeling framework helps to overcome this challenge by assuming the investment decisions only occur at three points in time. Each optimization period is performed independently, making the model myopic - current investment decisions are assumed to be made under the assumption that no future knowledge of the system is known. This is done to help account for the non-deterministic nature of the the power system, investment decisions are made and capacities are installed without knowing the operational situations the system will encounter.

Given time and computation constraints, the demand is modeled to be inelastic in the investment optimization and is run as a cost minimization optimization. In addition, given the mathematical formulation of the optimization of the MGA optimization, the demand is modeled to be inelastic in the MGA analysis. The investment and MGA optimization are solving with the constraint of having to exactly satisfy the demand at each node for every point in time. To make the model more closely represent reality and generate more realistic electricity prices, the demand is assumed to be elastic in the operations optimization and therefore, the operations optimization is modeled as a welfare maximization optimization problem. The details of each component of the modeling framework are described in further detail in the following sections.

# 3.3. Investment optimization

The investment optimization model developed is a myopic partial-equilibrium cost minimization model, investors do not have perfect foresight, and therefore the market modeled deviates from a perfect competitive energy market. The modeled market is assumed to be competitive and follow marginal-cost pricing. The demand is modeled as being elastic, the demand changes in response to changes in the price. The demand is modeled to be elastic to provide a more accurate representation of the real electricity market. The investment optimization is formulated as a welfare maximization optimization, with the objective to maximize the total economic welfare.

#### **Objective function**

The optimization is formulate as a welfare maximization optimization. For the investment optimization, the total welfare is maximized to find the long-term cost-optimal electricity system. The welfare is determined by maximizing the customer utility subtracted by the total annual system cost. The total annual system costs is the sum of the annualized investment costs (the annualized capital costs for generation, storage, and transmission capacities) and the annual operational costs for the snapshots of time identified in the model are minimized. The results of the optimization problem give the optimal set of generator, storage, and transmission capacities that when operated optimally result in the lowest overall costs for the optimal demand determined by the model.

$$Maximize \begin{pmatrix} \text{Total economic} \\ \text{welfare} \end{pmatrix} = \sum \begin{pmatrix} \text{Consumer utility} \\ (\text{value function}) \end{pmatrix} - \sum \begin{pmatrix} \text{Generator costs} \\ (\text{supply function}) \end{pmatrix}$$
(3.1)

The optimization function maximizes the total economic welfare by maximizing the consumer utility subtracted by the generation costs. The optimization is run for one representative year with multiple time steps to represent the variety of weather and demand conditions that exist throughout the year. The detailed objective function is shown in 3.2 below. The objective function and constraints are adapted from (Neumann and Brown, 2019).

$$\max_{K,H,F,k,h} \left[ \sum_{n,t} (U_{n,t}d_{n,t}) - \sum_{n,g} (c_{n,g}K_{n,g}) + \sum_{n,s} (c_{n,s}H_{n,s}) + \sum_{l} (c_{l}F_{l}) + \sum_{n,g,t} (w_{t}o_{n,g}k_{n,g,t}) + \sum_{n,s,t} (w_{t}o_{n,s}(h_{n,s,t}^{-} + h_{n,s,t}^{+})) \right]$$

$$(3.2)$$

The objective function maximizes the utility of the consumer  $U_{n,t}$  at each node n for each time t multiplied by demand at the respective node at time t,  $d_{n,t}$  subtracted by the total system costs. The total system costs consist of the generator capacities  $K_{n,g}$  at each node n for each generator technology g multiplied by their annualized capital cost  $c_{n,g}$ , the storage capacities  $H_{n,s}$  at each node n for each generator technology g multiplied by their annualized capital cost  $c_{n,g}$ , the storage capacities  $H_{n,s}$  at each node n for each line l multiplied by their annualized capital cost  $c_l$ , dispatch of each generator technology  $k_{n,g,t}$  at node n multiplied by their operating costs  $o_{n,g}$  and the time step weight  $w_t$ , and dispatch of each storage technology  $h_{n,s,t}^{-/+}$  at node n multiplied by their operating costs  $o_{n,g}$  and the time step weight  $w_t$ . Each time period is given a weighting,  $w_t$ . The weightings are chosen such that the sum of the weights over the chosen time steps t equal 8,760, representing a full year of operation. The optimization function minimizes total system cost per year. For simplicity, start-up and shut-down costs of generators is not included in the objective function.

Only capital cost of generation, storage, and transmission capacities installed in the period being optimized are included in the objective function. Previously installed capacities are considered sunk cost and therefore, their capital cost is not included in the objective function.

Solving the investment optimization gives the optimal power system configuration. In addition, the objective function gives the optimal system cost, which is then used in the MGA optimization, described in the next section.

#### Elastic demand

In the case of the elastic demand, the consumer utility function can be determined using demand curves. Elastic demand is incorporated into ESOMs to align more closely with real world electricity

market observations (Decarolis et al., 2017). Incorporating elastic demand in ESOMs, allows electricity demand to be responsive to price changes such that an increase in price leads to a decrease in demand and an decrease in price leads to an increase in demand. Linearized demand curves are used to represent how demand changes with a change in electricity price.

For the case where demand is assumed to be elastic, utility of the consumers needs to be calculated. To begin with, the coefficients for demand elasticity need to be determined. A reference demand  $D_{ref}$ , reference price  $P_{ref}$ , and the elasticity of demand  $b^{-1}$  are calibrated to historical price and demand values for the respective market being modeled. Using these values the necessary parameters can be determined to formulate the utility function for consumers, U(d). The utility function or total benefits is given by the area under the demand curve bounded by the equilibrium price point (point that gives the demand and market price for a given time step), the point the supply and demand curves intersect as is shown in Figure 3.3.



Figure 3.3: Supply and demand curves. The dashed triangle is the utility.

Therefore, to determine the utility, the demand function for each time step modeled must be formulated. The demand function is:

$$p = P_0 - bd \tag{3.3}$$

where p is the price,  $P_0$  is the price when demand equals zero (the y-intercept of the demand curve), d is the quantity demanded, and b is the inverse of the price elasticity of demand (the slope of the demand curve). The price elasticity of demand is how much the quantity is demanded given a change in price or in other words, the ratio of the percentage change in demand to the percentage change in price.

The demand curve for each time step should be unique and it should be correlated with the demand hourly time series fed into the model. The elasticity of demand is assumed to remain constant from one time step to the next and thus, the slope of the demand curve remains constant. Given the original demand for the particular time step  $D_{org}$ , the elasticity of demand  $b^{-1}$  and the assumed reference price,  $P_{ref}$ , the y-intercept,  $P_0$  is calculated for each time step with the demand equation:

$$P_0(t) = bD_{org}(t) + P_{ref}$$
(3.4)

These parameters and utility function given below are then used to determine the total benefits.

$$U_{n,b,t}(d_{n,b,t}) = (P_0 + 0.5bd_{n,b,t})d_{n,b,t} \qquad \forall (n,t)$$
(3.5)

where  $P_0$  is the price when demand equals zero (the y-intercept of the demand curve), d is the demand, and b is the inverse of the price elasticity of demand. The price elasticity of demand is how much the quantity is demanded given a change in price or in other words, the ratio of the percentage change in demand to the percentage change in price.

#### 3.3.1. Modeling to generate alternatives (MGA) optimization

The MGA analysis is used to determine the range of near-optimal solutions. The MGA methodology used in this research is modeled after (Neumann and Brown, 2019). The optimal system cost determined in the investment optimization is used to define a new constraint. The optimal solution value ( $f^*$ ) plus an acceptable relative cost increase ( $\epsilon$ ) is then used to constrain the original feasible space as a new constraint in the optimization problems to explore the near-optimal feasible space.

$$\min f(K,H,F,k,h) \le (1+\epsilon) * f^*$$
(3.6)

The chosen epsilon value (slack value) is up to the discretion of the modeler. Several previous MGA studies use a range of epsilon values to examine the affect that epsilon has on the final output (Neumann and Brown, 2019; DeCarolis et al., 2016). in this study, an epsilon value of 10% is chosen. It is important to note that a medium-term optimization is used in this study and therefore, the investment optimization performed for the optimal solution minimizes the costs of additional capacity that is added to the system. The cost of existing technology is assumed to be sunk cost and is therefore not included in the total system costs. Since the optimal cost is used as the basis of the cost constraint for the MGA analysis is within epsilon of the cost of additional technology required in the system per investment period.

The new objective function becomes the minimization or maximization of the sums of subsets of generation, storage and transmission capacity expansion subject to the new allowable cost increase constraint.

$$min(x) \text{ or } max(x) \tag{3.7}$$

Where x, the decision variable, is the subset of generation, storage, or transmission capacity.



The set of new optimization problems are solved to determine the range of near optimal solutions. The resulting power system configurations generated by the investment optimization and MGA optimization



are input into the operations optimization and are run for a range of weather and demand years. The next section details the operations optimization.

#### 3.3.2. Operations optimization

Similar to the investment optimization, the operations optimization is modeled as a welfare maximization problem. For short-term equilibrium in the operations optimization, the objective function only maximizes over the short-term costs, which is the difference between the consumer utility and the producer operation costs. The capital costs (fixed costs) are excluded and the objective function is:

$$\max_{d_{n,t},k_{n,g,t},h_{n,s,t}} \left[ \sum_{n,b,t} U_{n,t}(d_{n,t}) - \left( \sum_{n,g,t} (w_t o_{n,g} k_{n,g,t}) + \sum_{n,s,t} o_{n,s,t} (h_{n,s,t}^- + h_{n,s,t}^+) \right) \right]$$
(3.8)

The objective function maximizes the utility of the consumer  $U_{n,t}$  at each node *n* for each time *t* multiplied by demand at the respective node at time *t*,  $d_{n,t}$  subtracted by the system operating costs. The system operating costs described in detail in the investment optimization section.

$$\max_{k_{n,g,t},h_{n,s,t}} \sum_{n,g,t} (w_t o_{n,g} k_{n,g,t}) + \sum_{n,s,t} o_{n,s,t} (h_{n,s,t}^- + h_{n,s,t}^+)$$

### 3.4. Optimization model constraints

The following section provides the constraints for all optimization models used in the modeling framework.

#### 3.4.1. Generator constraints

To solve the optimization function, the objective function is subjected to several constraints. The dispatch of generators  $k_{n,g,t}$  are constrained by the generator capacity  $K_{n,g}$  and the minimum and maximum time variable availability of the generator,  $\tilde{k}_{n,g,t}$  and  $\bar{k}_{n,g,t}$  respectively. The time variable availability of the generator is given per unit of installed capacity,  $K_{n,g}$ . The minimum time variable availability  $\tilde{k}_{n,g,t}$  of the generator signifies the lower bound of the installed capacity that must be in operation for the given time. The maximum time variable availability  $\bar{k}_{n,g,t}$  of the generator signifies the upper bound of the installed capacity that can be in operation for the given time.

$$\tilde{k}_{n,g,t} \le k_{n,g,t} \le \tilde{k}_{n,g} K_{n,g} \qquad \forall n, g, t$$
(3.10)

Conventional generators (coal, gas, and nuclear) are assumed to be fully flexible -  $\tilde{k}_{n,g,t}$  is 0 and  $\bar{k}_{n,g,t}$  is based on a randomized parameter to model times when the generator unit is offline for maintenance or for other non planned reasons. Therefore, for conventional generators, 3.10 becomes:

$$0 \le k_{n,g,t} \le K_{n,g} \qquad \forall n, g, t \tag{3.11}$$

For VREs, 3.10 becomes:

$$0 \le k_{n,g,t} \le k_{n,g,t} K_{n,g} \qquad \forall n, g, t$$
(3.12)

where  $\bar{k}_{n,a,t}$  is the weather dependent power availability of the VRE.

For the investment optimization and the MGA, the installed capacity of generators are optimized within bounds of minimum and maximum installable potential values  $K_{n,g}^{min}$  and  $K_{n,g}^{max}$ , respectively.

$$K_{n,g}^{\min} \le K_{n,g} \le K_{n,g}^{\max} \qquad \forall n,g$$
(3.13)

The capacity bounds are determined by existing/previously installed capacities, governmental phaseout decommissioning plans, or maximum renewable installation potential. For the first optimization period, 2030, currently existing installed capacities are used to define the minimum bounds  $K_{n,g}^{min}$ ,  $H_{n,s}^{min}$ , and  $F_l^{min}$ . For the consecutive optimization periods, the minimum capacities are the optimal capacity from the previous optimization period. The maximum capacities for the conventional generators are determined from governmental decommissioning plans and the maximum capacities for VREs are the maximum renewable installation potentials, given in Table A.2.

#### 3.4.2. Storage unit constraints

Similar to dispatch constraint for generators given in 3.10, the charging and discharging of storage units  $h_{n,s,t}^-$  is constrained by the storage power capacity  $H_{n,s}$ :

$$0 \le h_{n,s,t}^+ \le H_{n,s}^+ \quad \forall n, s, t$$
 (3.14)

$$0 \le h_{n,s,t}^- \le H_{n,s}^- \quad \forall n, s, t$$
 (3.15)

In addition, the state of charge of the storage unit  $soc_{n,s,t}$  is constrained by the nominal power,  $H_{n,s}$  multiplied by the number of hours that are required to fill the storage unit to the maximum state of charge,  $r_{n,s}$ . In this research, the number of hours required to fill a storage unit to maximum charge are set to 6 hours for batteries and 168 hours for hydrogen storage.

$$0 \le soc_{n,s,t} \le H_{n,s} * r_{n,s} \qquad \forall n, s, t$$
(3.16)

The state of charge  $soc_{n.s.t}$  has to be consistent from one time step to the next and therefore,

$$soc_{n,s,t} = soc_{n,s,t-1} + \eta_{n,s}^+ h_{n,s,t}^+ - \frac{1}{\eta_{n,s}^-} h_{n,s,t}^- \quad \forall n, s, t$$
 (3.17)

The state of charge must equal the state of charge at the previous time step  $soc_{n,s,t-1}$  plus the amount of power charged to the batter (the efficiency of charging  $\eta_{n,s}^+$  multiplied by the power charged  $h_{n,s,t}^+$ ) minus the amount of power discharge from the battery (the amount of power discharged  $h_{n,s,t}^-$  divided by the efficiency of discharging  $\eta_{n,s}^-$ . For simplification, it is assumed that the storage units have no standing losses (self-discharging leakage rate). For the investment optimization and the MGA, the installed capacity storage units are optimized within bounds of minimum and maximum installable potential values,  $H_{n,s}^{min}$  and  $H_{n,s}^{max}$ , respectively.

$$H_{n,s}^{\min} \le H_{n,s} \le H_{n,s}^{\max} \qquad \forall n,s$$
(3.18)

For the first investment optimization period, 2030, the minimum bounds are the currently installed battery or hydrogen storage. The maximum bound is infinity. For the consecutive optimization periods, the minimum bound is the optimal storage unit capacity from the previous optimization period and the maximum bound remains infinity.

#### 3.4.3. Transmission constraints

The flow in all transmission lines  $f_{n,t}$  are constrained by their capacities  $F_l$ .

$$|f_{n,t}| \le F_l \qquad \forall \ l,t \tag{3.19}$$

For the investment optimization and the MGA, the installed capacity of transmission are optimized within bounds of minimum and maximum installable potential values,  $F_l^{min}$  and  $F_l^{max}$ , respectively.

$$F_l^{min} \le F_l \le F_l^{max} \quad \forall l \tag{3.20}$$

For the first investment optimization period, 2030, the minimum transmission capacity bound is the currently installed transmission. The maximum bound is infinity. For the consecutive optimization periods, the minimum bound is the optimal storage unit capacity from the previous optimization period and the maximum bound remains infinity. The transmission lines are modeled as lossless.

#### 3.4.4. Nodal energy balance

For each point in time the demand at each node n must be exactly satisfied by the energy generated by the generators at node n  $k_{n,g,t}$ , the discharge of storage units at node n  $h_{n,s,t}^-$ , minus the charging of storage units at node n  $h_{n,s,t}^+$ , and the flow from the transmission lines to node n,  $f_{l,t}$ . This gives the nodal balance constraint detailed below.

$$\sum_{g} k_{n,g,t} + \sum_{s} (h_{n,s,t}^{-} - h_{n,s,t}^{+}) + \sum_{l} (\alpha_{l,n,t} f_{l,t}) = d_{n,t} \leftrightarrow \lambda_{n,t} \quad \forall n, t$$

$$\alpha_{l,n,t} : -1 \text{ if } l \text{ starts at } n, \ l \text{ withdraws power from } i$$

$$\alpha_{l,n,t} : 1 \text{ if } l \text{ starts at } n, \ l \text{ supplies power from } i$$
(3.21)

The shadow price (i.e. dual variable) of the nodal energy balance gives  $\lambda_{n,t}$ , the marginal price at each bus for each period of time modeled.

#### 3.4.5. Emission constraint

An emissions limit  $CAP_{CO_2}$  can be imposed on the system as a global constraint. The emissions can be constrained by calculating the sum of emissions for each generator over the course of the year modelled. The emissions per generator are calculated using the carbon intensities of the fuel used in the generator  $e_g$  and the efficiency of the generator  $\eta_{n,g}$ :

$$\sum_{n,g,t} w_t \frac{1}{\eta_{n,g}} e_g k_{n,g,t} \le CAP_{CO_2} \leftrightarrow \mu_{CO_2}$$
(3.22)

 $\mu CO_2$  is the shadow price of the CO2 emissions and therefore, identifies the CO2 price that is necessary to reach the carbon emission limit specified in the constraint.

#### 3.4.6. Investment calculation

Given the electricity prices and the generator and storage unit dispatch determined in the operations optimization of all generated alternative power systems, the NPV is calculated for each technology. Net

present value (NPV) is a basic financial calculation to assess the value of a project. The NPV is the sum of the discounted cash flows, costs and revenues, with a certain interest rate, r, over the assumed lifespan of the asset (Petitet, 2017).

Net present value (NPV) (Brown, 2020):

$$NPV = \sum_{y=0}^{Y} \frac{-C_y - O_y + R_y}{(1+r)^y}$$
(3.23)

 $C_{y}$ : capital expenditure in year y

 $O_{v}$  : operating & maintenance expenditure in year y

 $R_{y}$  : revenue in year y

r : discount rate

If the NPV is positive, the investment is economically profitable and indicates that the investment is worthwhile. Whereas, a negative NPV indicates that the project should be rejected (Petitet, 2017). To account for the fact that t

Given that the NPV is calculated using a discount rate, the order of cash flow is significant; making earlier profits more desirable. To account for this and the randomness of weather and demand data for each operation year, bootstrapping, a statistical technique using random sampling with replacement, is performed over the operation optimization results. The set of dispatch and electricity price results from the operations optimization are randomly selected 1000 times and used to calculate 1000 different NPVs for each full model run. This provides a range of possible NPV outcomes.

# 4

# Case Study

A simplified optimization model is built to practically show the modeling framework described in Chapter 3 for analyzing the business case of VREs and storage using ESOMS as the power system evolves to meet emission reduction targets. The case study explores the use of a simplified energy optimization model to analyze investments in the power system from the perspective of an energy generation company. The case study's objective is to provide a practical example of the modeling framework developed described in Chapter 4 and to gain insights into the profitability of generation and storage technologies installed in 2030 in the Netherlands.

## 4.1. Case study details

The model for the case study is built in Python with the open-source optimization package, Pyomo, and the CPLEX optimization solver. The case study considers a two-node, electricity only market. Following the methods described in Chapter 4, a multi-stage, medium-term optimization is performed, where each investment period is optimized independently. Following the investment optimization, an MGA analysis is performed on each of the investment periods. Using the optimal and MGA alternative power system configurations, 10 operations optimizations are performed. The model only considers the Netherlands and Germany in a two node system. A spatial resolution of two nodes was chosen as this allows for the simplest model while still being able to model all system components, including transmission. This allows for the case study to reveal how the developed modeling framework can be applied and to allow for a clearer understanding of the effect system components have on the overall model results. Germany was chosen as the second node, as it is the country the Netherlands shares the largest interconnection capacity with. The generation technologies modeled are solar, onshore wind, offshore wind, CCGT, OCGT, coal, lignite, biomass, and nuclear. The storage technologies included in the model are batteries and hydrogen storage. Minimum generation capacities in 2030 are assumed to be current installed capacities in each respective country (based on 2020 installed capacities). The time frame of the model is from 2030-2060, with investment years 2030, 2040, and 2050 (installation and decommissioning of capacities are assumed to only occur in these years). It is assumed that the demand and installed capacity remain constant throughout each 10 year investment segment (20302039, 2040-2049, 2050-2059). The decommissioning of generation capacity is assumed to follow governmental decommissioning plans or plant lifetimes. The maximum VREs capacities are bounded by the potential renewable energy capacities for each respective country. Demand is modeled to follow historical data, scaled to represent increasing demand over time. The demand factor is consistent across each 10-year investment period and increases by 15% per ten-year period. It is assumed that installed capacity remains constant throughout each 10 year investment segment (2030-2039, 2040-2049, 2050-2059). A greenhouse gas emissions constraint is included in the model to follow governmental  $CO_2eq$  reduction targets. The greenhouse gas emission reduction targets outlined by the Dutch and German governments are assumed to be achieved.

The model assumes that conventional generators are fully flexible and therefore, their dispatch power is only limited by the capacity of the generator type. In addition, generators are assumed to have no startup or shutdown time. The operation costs for storage units is assumed to be zero. The capacity optimization is a medium-term optimization in terms of the capital stock. The optimization uses the existing installed capacities as a minimum for the initial investment period, 2030 and uses the previously determined capital stock as the minimum for the next investment optimization. All existing capacity is assumed to be sunk cost and is therefore excluded from the investment optimization calculation. The investments made in 2030 are assessed. Therefore, the NPVs calculated are for investments in 2030 and are normalized to 1 MW of capacity for each technology. The NPVs are calculated using a 7% discount rate.

# 4.2. Experimental setup

The experimental set up follows the modeling framework explained in Section 3.2 and the model flow diagram shown in Figure 3.1. A step by step explanation of the experimental setup is given below. See Figure 4.1 for a graphical representation of the experimental setup. This figure represents the setup for the optimal run and each MGA run (minimizing and maximizing each technology) and is therefore repeated 19 times.



Figure 4.1: Graphical representation of the set-up for the optimal run and each MGA run.

1. A brownfield investment optimization for each investment period, 2030, 2040, 2050 is performed to determine the optimal installed capacity for each respective year. These optimizations are

based on historical weather and demand data from 2013. Existing capacity or capacity built in the previous optimization period is given. Existing capacities are considered sunk cost and therefore are disregarded in the optimization. The optimization determines endogenous investments in new capital stock.

- 2. MGA is performed using the optimal cost determined in step 1 for each of the investment optimization periods. The optimal cost plus a 10% cost deviation is encoded as a constraint. The objective function becomes the minimization or maximization of a certain technology. Alternatives are generated by minimizing and maximizing each technology (generators, storage units, transmission) one by one and determining the set of installed capacities while remaining within 10% of the cost optimal solution. 18 alternatives will be generated for each investment year (2030, 2040, 2050). Therefore, a total of 54 MGA optimizations will be performed. Similar to the optimal investment optimization, the MGA optimization is a multi-stage investment, where each investment year is optimized independently. The storage, generation, and transmission portfolios determined in the previous optimization period for the MGA run with the same objective will be used as minimum capacities for the the consecutive run.
- 3. A short-term optimization (dispatch optimization, only optimizing operations) is performed using the installed capacities determined in each of the MGA alternatives for each of the years in the investment periods (2030-2039, 2040-2049, 2050-2059). A set of 10 years of different historical demand, wind and solar capacity factor data (historical data from 2007-2017) are used to run 10 different short-term optimizations, each representing one year. The results give a range of future electricity prices and generator dispatch scenarios.
- 4. The resulting electricity prices and generator dispatch are used to determine the market value and NPV of each respective technology.
- 5. To account for the randomness of operation years, bootstrapping with replacement is performed over the operation optimization results. The dispatch and electricity price results from the operations optimization are randomly selected 1000 times and used to calculate 1000 different NPVs for each full model run. This provides a range of 1000 possible NPVs per full model run.

# 4.3. Model data

Techno-economic input data used in the model includes capital costs, variable and fixed operation costs, technology efficiency, technology lifetime, fuel carbon intensity, fuel price, greenhouse gas emissions, and potential renewable capacity. The majority of the data used in the model comes from the open source model, Python for Power System Analysis (PyPSA) (Brown, Hörsch, and Schlachtberger, 2018). The greenhouse gas emissions used to set the emission constraint in the model are from established government targets for the Netherlands and Germany of the allowable emissions from the power sector for the respective years. The VRE capacities for the Netherlands and Germany are the maximum installed capacity given geographical constraints. The historical installed generator capacity in the Netherlands and Germany are used as a basis for the model and to verify the model. For a complete table of all input parameters see Appendix A.

The demand is based on historical demand data from 2007-2017 for the Netherlands and Germany. To account for the anticipated increase in demand over the coming decades, the demand is increased by an additional 15% per investment stage relative to the reference year, 2020. In addition, the solar

and wind capacity factors are based on historical capacity factors from the Netherlands and Germany for the years 2007-2017 (Pfenninger and Staffell, 2016; Staffell and Pfenninger, 2016). All monetary values used in this study are in 2015 Euros.

# 4.4. Elastic demand parameters

The elastic demand parameters are calibrated using historical data. In this case study, the demand and weather data from 2015 is used to calibrate the elasticity demand parameters. The historical capital stock for the Netherlands and Germany are given as a parameter to the model. The installed generator capacities for the Netherlands, Germany, and total for both the Netherlands and Germany for 2015 are shown in Figure 4.2 below.



Figure 4.2: Bar chart of installed capacities by generator and storage types for both the Netherlands and Germany for 2015.

Many iterations of operations optimizations with varying elastic demand coefficients are then performed with the given historical capacities and weather and demand data to determine the elastic demand coefficients to be used in the model. The reference price ( $P_{ref}$ ) is set at the average electricity price in 2015,  $41 \in /MWh$  for the Netherlands and  $37 \in /MWh$  for Germany. There are several model limitations that lead to the inability to accurately calibrate the elastic demand coefficients to historical values, including the exclusion of start-up and shut-down costs, ancillary services, must-run generators, strategic market bidding, or unique generators for each technology type (each technology is modeled as one generator with one set of marginal and capital costs). The average price and the price duration curves of the model are compared to historical values and the deviation of model prices from historical values are used to calibrate the elasticity coefficients. The slope of the demand curve, b, is calibrated to be approximately 0.01.

The price duration curve shows the electricity price over time, plotted in order of decreasing magnitude, revealing the number of hours in a year the electricity price is a given value. The load duration curves for the 2015 historical values and the 2015 model values are shown in Figure 4.3 below.

The most apparent difference between the historical prices and the model prices is the smoothness of the historical price duration curve compared to the model prices. This difference is primarily due to



Figure 4.3: Price duration curve for the 2015 historical prices and the 2015 model prices.

modeling each generator technology as one unit, leading to the marginal cost to be the same for all electricity generation of the same technologies. As the marginal cost of the marginal generator sets the electricity price, this leads to the electricity price being the same price for many hours thorough out the year. The average electricity price in both the historical prices and the model prices is  $40 \in /MWh$ .

5

# Model verification & validation

The model is verified and validated to ensure it can be used to adequately represent the profitability of investments in the power system, in other words, that the model is appropriate for the purpose it is designed for. Model verification and validation are the primary processes to provide evidence that the developed model can be used for its intended purposes (Thacker et al., 2004). Model verification is done to ensure that the computer programmed model correctly represents the conceptual mathematical model and accomplishes the conceptual goals outlined in Chapter 3 and 4. Model validation is the process of determining if the constructed model represents the real world to an acceptable degree. The model was consistently verified and validated during the development phase to ensure that each additional component added to the model was modeled correctly.

## 5.1. Model verification

Throughout the process of developing the model, there were numerous verification check points. The model was developed in a step by step manner, where in each step of the development a working model is built and each consecutive step, additional levels of complexity are added to the model, beginning with the most simplistic working model. A very simplistic one node model, only considering conventional generators, was built at the beginning. This base model could be easily verified against the graphical solution for the conceptual model. After this model was verified, additional more complex components were added and tested one by one to ensure that they were coded correctly in the model. These additional components include an additional node, renewable generators, storage units, the MGA optimization, minimum installed capacity for existing capital, governmental decommissioning targets, and constraints for maximum installable capacity for renewable generators.

### 5.1.1. MGA verification

The MGA optimization is verified by ensuring that for each MGA optimization, the total cost of the optimized alternative scenario (scenario minimizing or maximizing a certain technology) is within the epsilon value of the total optimal system cost. This analysis is performed to check whether or not

the computer simulation model developed aligns with the conceptual definition of the MGA analysis, defined in Section 4.3.

## 5.2. Model validation

#### 5.2.1. Capacity optimization validation

To validate the model, the final complete model is compared to real world historical data. To first test the capacity optimization, the optimal capacity optimization and MGA optimization are run for 2020 and the resulting capital stock is compared to the actual installed capacity in 2020 for the Netherlands and Germany. This validation is done to determine if the real installed capacity in 2020 are within the range of generated electricity systems using the methodology developed in Chapter 4.

This validation process supports the importance of performing an MGA analysis when performing energy system optimization modeling, as described in Section 2.2. The real power system has been shown to deviate from the optimal system due to the other, non-cost, factors (i.e social acceptance, risk of investments, imperfection of the market, irrational actors, decentralized decision making) that affect which energy system gets developed (Trutnevyte, 2016). The validation using the methodology developed in Chapters 3 and 4 and is run with precisely the same data input as the full working model. The demand and weather year is based on 2013 historical data, the same data year used in capacity optimization phase. Following the developed methodology, the capacity optimization is a medium-term optimization and therefore, all existing capacity is included. Per the methodology, the model assumes ten year investment periods, only allowing investments to occur in the first year of the investment period and takes the existing capacity to be the capacity installed 10-years earlier. Therefore, to test the year 2020, 2010 was used as the base year. All installed capacity from 2010, excluding decommissioned generators, are fed into the model as minimum capacities values per technology. Therefore, model includes the existing installed capacity and is able to optimize the system by adding additional capacity in addition to the existing installed capacity. Figure 5.1 compares the optimal and all MGA optimization solutions to the actual historical installed capacity in 2020. The figure reveals that the historical installed capacity values for each generation and storage type are within the range of installed capacities generated by the model, validating the models use to represent the real world. This validation process reveals that the range of solutions generated using the MGA methodology can encompasses the real world energy transition. This form of ex-post modeling can help reveal the necessity of uncertainty techniques when modeling the future energy system. As is revealed in Figure 5.1, the real world power system configuration deviates significantly from the optimal solution but the real world system falls within the range of MGA alternative system results.

It is also important to ensure that the build-out capacities from the capacity optimization in the model align with the historical values for each country. Due to the nature of the optimization model, capacity is built in the most cost-effective way, not considering individual country's policies or desire for energy independence. In the real world, a country's individual decision-making on power system developments can be a cause of deviation from the overall cost optimal solution. In addition, innate to the design of the model and due to the difficulty of associating emissions to each respective country, because of cross-boarder transmission, the  $CO_2e$  constraint in the model is the cumulative allowable emissions for the Netherlands and Germany. In reality, each country has their own emission reduction targets that they aim to achieve and therefore, the transition in each country might deviate from the model in



Figure 5.1: Bar chart of installed capacities by generator and storage types for both the Netherlands and Germany of the historical, optimal, and each MGA alternative.

this regard. The model does not consider that each countries power system will transition in light of achieving their emission reduction targets while maintaining a certain level of domestically produced power for energy security. The model determines the (near) cost-optimal without considering if that solution would achieve each respective countries power generation emission reduction targets and ensure each country retains a certain amount of domestically produced electricity. This could be a source of deviation of the model from reality. The EU as a whole has emission reduction targets (i.e. the European Green Deal's 55% emission reduction target by 2030 and net zero emissions by 2050). These generalized targets are inline with the way the model is currently programmed.

#### 5.2.2. Insights from capacity optimization validation results

Insights can be drawn from the validation analysis to better understand how the model represents real world installed capacities. As can be seen from Figure 5.1, the actual cumulative installed generation capacities in 2020 are greater than the majority of the MGA alternatives and has approximately 29 GW (or 16%) more installed capacity than the cost optimal solution. It is important to acknowledge that the model is optimized on the basis of 2013 weather and demand data and therefore one of the sources of the deviation of the resulting optimal model from the actual cost optimal power system for 2020 is the difference between historical demand and weather data for that year and those of 2013. This is representative of a main source of uncertainty when optimizing the future energy system, the weather and demand data. The parametric uncertainty is outside the scope of this analysis but should be explored in further research, particularly how parametric uncertainty can be explore with the MGA uncertainty technique.

### 5.2.3. Operations optimization & electricity price validation

The full model is validated, the capacity optimization and the operations optimization, by performing an ex-post optimization of 2019. Similar to the capacity optimization validation performed in the previous section, 2010 is used as the base year. The operations optimization and electricity price validation is done for the year 2019, as historical electricity data for the whole year was available at the time of

performing the analysis. Therefore the validation of the electricity prices is done assuming the base year is 2010 and the capacity and operations optimization is performed for 9 years later, the year 2019. This deviates from the actual model, as the actual model calculates the capacity optimization in 10 year segments, not 9 years. For this reason, the capacity optimization is validated for 2020 to account for a 10 year optimization segment.



Figure 5.2: Bar chart of installed capacities by generator and storage types for both the Netherlands and Germany of the 2010 and 2019 historical values and the 2019 optimal values and values for each MGA run from the model.

The resulting electricity prices for each alternative is compared to historical values. These price duration comparison graphs are in Appendix C. The comparison of the model derived price duration curves with the historical values reveals that the model typically has higher electricity prices than occurred in reality. However, the average electricity prices are encompassed in the set of alternative runs. This provides verification that the set of near optimal solutions can be used to explore potential electricity prices. The higher electricity prices in the model can be due to a number of factors, primarily that the input parameters do not precisely align with real world values. These values include the cost data, technology data, and weather and demand data. This suggests that a parametric uncertainty analysis should be incorporated in future research to account for the uncertainty in these parameters.

#### 5.2.4. Insights from operations optimization validation

There are larger variations in real world electricity prices compared to the model generated electricity price. This is a result of the model simplifications compared to the real world power system. First of all, each technology type is model as one larger unit, instead of several different unique unit for each type of technology as it is in the real world. The different units in reality cause differences in the marginal cost of each plant, leading to slightly different bid prices into the electricity market. These different bid prices leads to more variability in electricity price than is experienced in the model. In addition, the model does not consider shut-down or start-up costs, must-run generators, or market behavior of system actors. This leads to no negative prices in the model even though negative prices are possible and seen in the real power market. To obtain a model that more accurately represents the real power system, the additional components should be added to the model. The model only considers a two-node system, therefore, excluding the effects that the other interconnected countries have on the Dutch and Germany electricity systems and market. The Netherlands has additional direct electric transmission

lines with Belgium, Great Britain, Norway, and Denmark. The interconnection between countries in Europe can have a large affect on individual countries electricity prices. Therefore, excluding these interconnection capacities in the model can be another cause of the deviation in the model electricity prices from historical values.

In addition, due to the simplifications of the model, demand elasticity parameters are difficult to model to historical values. A more complex model and proper calibration of demand parameters with historical prices, would lead to a more accurate representation of the real world electricity prices and should be performed in future work.

# 6

# Case study: Results & analysis

# 6.1. Optimal results

In the first phase of the problem, the long-term investment optimization is performed to generate the optimal solution of installed capacities for 2030, 2040, and 2050. The optimal total installed capacities for the modeled 2-node system are given in Figure 6.1.



Figure 6.1: Bar chart comparing the installed total generator, storage, and transmission capacities for both the Netherlands and Germany for the optimal solution in 2030, 2040, and 2050. The generator and storage capacities are shown by the bars (left axis) and the transmission capacities are given with the points on the graph (right axis).

As can be seen from the figure, the necessary installed capacities increase over the 3 investment periods. This increase stems from a combination of the increase in demand that is assumed in the model and the increase in capacity that is required to account for the intermittency of VREs. In the 2050 time frame, wind dominates the system with solar providing the remainder of necessary generation capacity. The optimal transmission capacity is 32 GW, approximately 7 times the currently installed transmission between the Netherlands and Germany of 4.5 GW. In the zero emission scenario in 2050, the increased transmission capacity and the storage in the system help to smooth out the intermittency of the VREs. This is critical given that the only generation technologies in the system in 2050 are wind,

solar, and a little bit of biomass. To achieve these optimal installed capacities, the model endogenously determines a CO2 price of 69 €/MWh in 2030 and 170 €/MWh in 2040. In 2050, there is no CO2 price as CO2 emission constraint is set to zero, aligning with emission reduction targets. The CO2 price is the dual variable of the CO2 limit constraint in the investment optimization.

The total generation for the optimal 2-node system modeled for each of the investment optimization runs (2030, 2040, and 2050) are given in Figure 6.2, this is assuming the average weather and demand year that is used in the investment optimization, year 2013. Comparing Figure 6.2 to Figure 6.1, it is clear that total installed capacity experiences a more significant increase over the model time horizon than the total generation. The total consumption is modeled to increase an additional 15% per investment period from the base year, 2020. The consumption is approximately equal to generation in the 2030 optimal system. This results from the fact that no transmission losses are modeled and the system has no storage and therefore, all power generated is consumed. In 2040 and 2050, generation is greater than consumption due to the inclusion of storage in these years. Storage experiences losses, losses from battery charging and discharging and the production of hydrogen from electricity through electrolysis and back to electricity with a fuel cell.



Figure 6.2: Bar chart comparing the total generation for the Netherlands and Germany per technology for the optimal solution in 2030, 2040, 2050 of a average operations year (year used in the investment optimization).

For each of the investment periods, a sequential operations optimization is run 10 times, for a range of 10-years of historical weather and demand data. The capacities determined in the investment optimization for each respective investment period are used as input. The results give a variety of ten year sets of generation and electricity prices for each investment period. The variation in the generation per technology over these 10 years for each investment period for the optimization are shown in Figure 6.3.

The figure shows both the change in generation over each respective investment stage (2030-2039, 2040-49, 2050-59), as well as the variation in generation within each respective investment stage. The variation within the investment stages results from the variation in weather and demand data used for each of the short-term optimizations. Fossil production is nearly halved from 2030 to 2040, with no coal or lignite remaining in 2040, and further reduced to zero in the 2050 time frame. In the 2050-2059 time frame, the majority of the generation comes from the installed onshore wind.

Following the two step optimization of the optimal scenario, the MGA optimization is performed for each technology. The additional system costs of the optimal scenario plus a 10% cost increase are coded as a constraint. The optimization objective becomes the minimization or maximization of the



Figure 6.3: Bar chart comparing the total generation for both the Netherlands and Germany per technology for the optimal solution over the entire modeled time horizon: 2030-2059.

various generation, storage, and transmission technologies. The MGA optimization is run 16 times, minimizing and maximizing each technology. The resulting total installed capacities for the Netherlands and Germany for each MGA alternative as well as the optimal scenario are shown in Figure 6.4.



Figure 6.4: Bar chart comparing the capacities for both the Netherlands and Germany for each alternative generated using MGA for 2030, 2040, and 2050.

As a result of the MGA analysis, the figure reveals that within the 10% cost increase from the opti-

mal solution there are a wide variety of possible near optimal power systems. This provides important insights for both investors looking to invest in the future power market, as well as regulators trying to implement policy to push the deployment of clean technologies and achieve their greenhouse gas emission targets. There are many different near-optimal pathways that allow for emission reduction targets to be met. This opens up the possibilities and can provide support and an increased degree of freedom in decisions both for investors and for policy makers. The exploration and acknowledgement of these various pathways are essential because there are factors other than cost that affect the development of the power system, as explained in Section 2.2.6. These alternative factors are often difficult to incorporate or anticipate, but they can play an important role in the the development of the power system. Relaxing the cost constraint allows for the model to help encompass the range of energy systems that can occur in the reality. Figure 6.5 shows the range of installed capacities per technology for each investment period from the generated alternative power systems that are within 10% of the cost optimal solution.



Figure 6.5: Bar chart comparing the capacities for both the Netherlands and Germany for each alternative generated using MGA for 2030, 2040, and 2050. The grey plotted line markers denote the total maximum potential renewable energy capacities in the Netherlands and Germany.

As can be seen in Figure 6.5, several technologies have a wide range of possible installed capacities in the range of near-optimal solutions. It is important to note that in 2050, an increased amount of onshore wind, solar, and hydrogen storage are all required to be in the power system to remain within 10% of the cost optimal solution. Each of these technologies is required at capacities greater than their currently installed capacities. The greatest flexibility is installed onshore wind. Several conclusions can be made from figure 6.4 and 6.5. By 2040, the offshore wind and solar maximum alternatives reach the maximum potential for each respective technology in at least one MGA solution. By 2050, onshore wind reaches its maximum installed capacity in two of the near optimal solutions, the MGA run maximizing onshore wind capacity and the run minimizing offshore wind capacity. Onshore wind reaches its maximum potential in 11 near optimal solution runs and solar in 16 runs, all but the MGA run minimizing installed solar capacity.

# 6.2. Results for the Netherlands

The results for the Netherlands are given in this section. It is important to note that there is a global emission constraint that constrains the total emissions for both the Netherlands and Germany in the model. This causes the location of VRE technology development to be based on weather data for each respective location, within the limits of the built transmission capacity.

Figure 6.6 shows the optimal installed capacities for 2030, 2040, and 2050 in the Netherlands. Offshore wind experiences a significant increase between 2040 and 2050, dominating the power system in the 2050 time frame. This large increase in offshore wind is due to the fact that onshore wind reaches its maximum capacity potential in the Netherlands in 2030 and wind capacity factors are more favorable in the Netherlands compared to Germany. As there is no constraint set on how much electricity can be imported in each country in the model, the Netherlands' favorable wind capacity factor lead to a large installed capacity of wind and therefore, a large generation of wind. Therefore, to provide for the increase in demand in both the Netherlands and Germany and the power production losses from the decommissioning of gas plants, offshore wind is maximized in the Netherlands in 2050.



Figure 6.6: Bar chart comparing the total capacities the Netherlands for the optimal solution in 2030, 2040, and 2050.

Similar to Figure 6.4, Figure 6.7 shows the total installed capacities for the Netherlands for each MGA alternative and the optimal case for 2030, 2040, and 2050. As can be seen in the figure, there is a large amount of variation in the near-optimal solutions, particularly in the 2040 time frame. In 2040, offshore wind experiences the largest range of possible installed capacities, some alternatives experience no increase in offshore wind compared to the currently installed capacities, where as the offshore wind maximization scenario maximizes the possible installed capacity. By 2050, the majority of alternatives reach the maximum capacity potential for solar and wind. In 2050, offshore wind dominates the the Netherlands energy mix. These results suggest that there are many alternative transition pathways for the Netherlands, but many of the alternatives converge in order to reach full decarbonization in 2050, with much less variability of alternative solutions in the long-run. In most alternatives in 2040 and all but one alternative (minimizing onshore wind alternative) in 2050, onshore wind is maximized. Storage capacity in 2050 is comparably a larger portion of the overall installed capacity compared to storage in Germany.



Figure 6.7: Bar chart comparing the built out capacities in the Netherlands for each alternative generated using MGA for 2030, 2040, and 2050.

Figure 6.8 shows the range of installed capacities per technology for each investment period from the generated alternative power systems that are within 10% of the cost optimal solution. In some alternatives, the onshore wind is at the maximum potential for the Netherlands and by 2050. In all alternatives but one, minimizing the onshore wind, have the maximum possible capacity of onshore wind installed. This suggest that all onshore wind should be invested in and developed in the Netherlands.

To stay within 10% of the cost optimal solution, the model finds there needs to be between 3.9-44 GW of onshore wind, 1.7-151 GW of offshore wind, 6-46 GW of solar, 0-111 GW of battery storage, and 5-89 GW of hydrogen storage in the Netherlands for a carbon neutral power system in 2050. This is approximately 1-11 times the onshore wind capacity, 1-88 times the offshore wind capacity, and 1-8 times the solar capacity currently installed in the Netherlands. Substantial investments are required in the power system over the next 30 years to reach emission reduction targets and satisfy demand.

For each of the 51 power systems shown in Figure 6.7, 10 different optimizations are performed for the ten various demand and weather years. The operation optimization determines the optimal dispatch of each technology to satisfy the demand at the least cost. The optimization gives the generation of each technology at each hour and determines the LMP for each node at each hour. As described in Section 3.6.4, the LMP is the shadow price of the nodal energy balance constraint. Figure 6.9 shows a sample week of dispatch and electricity prices for the Netherlands and Germany from the model for the year 2030. The electricity prices at each hour are determined through the dual variables of the objective



Figure 6.8: Bar chart comparing the built out capacities in the Netherlands for each alternative generated using MGA for 2030, 2040, and 2050. The grey plotted line markers denote the total maximum potential renewable energy capacities in the Netherlands.

function (see Section 2.1.1). As can be seen from the graphs, at times when only VRE supplies all of the demand, the electricity price is close to  $0 \notin MWh$ , as the marginal price of VRE is assumed to be close to  $0 \notin MWh$ . The graphs show the variation of electricity due to the variation of which generator is the marginal generator at each hour. It is important to note that conventional generators marginal price includes the  $CO_2$  price. The peak electricity price around hour 5420 is a result of coal being the marginal generator and due to the high emissions from coal generation, the  $CO_2$  causes coal to have a high marginal price.



Figure 6.9: The top two graphs are stacked dispatch curves for all generation for a week in August from the 2030 model results. The bottom two graphs are the corresponding LMP curves over the same week in August. The graphs on the left are for the Netherlands and on the right are for Germany.

For each operations optimizations, a set of dispatch and marginal electricity prices for the Netherlands and Germany for every hour in a year are generated. In figure 6.10 histograms of the hourly electricity prices for each investment period are given. The electricity prices are clustered around certain prices. In reality, electricity prices are more distributed and experience more variation. In the model, each generator type is modeled as one large generator rather than each individual generators that have their own distinctive marginal price, as is the case in reality. This deviation from reality causes the electricity prices to be less distributed. In addition, the fact that only the Netherlands and Germany are modeled eliminates the influence of other surrounding countries on the electricity prices.



Figure 6.10: Histogram of all the LMP for all runs for each investment time period. Electricity prices greater than 200 EUR/MWh are all in the last bin.

For each investment period the electricity prices decease. In 2030, the average electricity price is €62/MWh, €66/MWh in 2040, and €64/MWh in 2050. Although the distribution of electricity prices changes through each of the investment periods modeled, the average electricity price remains around the same price. The increasing frequency of close to zero electricity prices over the time periods results from the increasing installed capacity of VRE sources in the system. With marginal based pricing, at times when there is enough VRE sources, the electricity price is set close to zero, as the marginal price of VREs is near zero. These lower electricity prices can lead to situations where the VREs cannibalize their own revenue. The absence of conventional generators means that CO2 pricing can not be used to ensure adequate electricity prices. However, the increase in storage capacity in the system can help to stabilize electricity prices through price arbitrage. Storage units can purchase electricity and charge or produce hydrogen at times when electricity is cheap and can sale electricity back to the grid (by discharging or producing electricity from hydrogen) at times when there is a shortage of renewable sources. This buying and selling of electricity by storage units helps to increase the variety of electricity prices rather than having a few instances throughout the year that have very high prices and the rest of the hours of the year have zero prices. The ability for storage to stable the electricity prices is limited by their storage capacities. In this optimization, the storage capacity of hydrogen and batteries is set to be that the batteries can discharge at full capacity for 6 hours and hydrogen for 168 hours.

The change in the distribution of electricity prices over the course of the year for each time period examined can be seen in Figure 6.11. The figure shows representative price duration curves for the optimal optimization solutions for one year in each of the investment periods (2030,2040, and 2050).



Figure 6.11: Price duration curves for 2030, 2040, and 2050 of the optimal scenario.

From the graph, it can be seen that with increasing investment periods, the number of hours that have close to zero electricity prices increases, correlated to increasing wind and solar generation. In addition, in the 2050 time frame, the increased capacity of storage in the system leads to a wide variety of electricity prices, as the storage units are able to buy and sell electricity. In future work, the electricity prices in a fully renewable system should be further explored. Allowing the storage unit capacity to be optimized could possibly lead to alternative electricity price behavior in the case of a fully renewable system. In addition, the energy system will progress to being more integrated; the future zero emission power system will be highly integrated with the rest of the energy system. Therefore, the other sectors of the energy system will influence the power system. However, hydrogen is used in other sectors and therefore has an associated commodity price. Hydrogen will likely be integrated through different energy systems and therefore will be sold at the set commodity price back into the power system. Future research should explore the integration of the power system with the entire energy system to get more accurate results.

The hourly electricity prices and dispatch of each respective technology are used to calculate the yearly revenue per technology. Bootstrapping is used to randomly select 10 years of revenue results within each investment period. This is done to help account for the uncertainty in future weather and demand, as each of the weather and demand years is equally as likely to occur in the future. Bootstrapping is performed 1,000 times and then the NPVs of each of the bootstrapping alternatives is calculated. The resulting NPVs for the Netherlands are shown in Figure 6.12. The NPVs are for capacity installed in 2030 and then run over the course of the next 30 years. The histogram provides the frequency of the NPVs For each installed technology. The color of the bars reveal the frequency and NPVs for each alternative optimization, the optimal and all MGA alternatives.

The majority of the NPVs for the generation technologies are clustered around 0, leading to no clearly favorable investments. Based on these results, investments in the Dutch power system in the 2030 time frame appear to have high risk. In a typical optimal result from a perfect foresight energy system optimization model, the NPVs of all generation and storage technologies in the system would have an NPV of zero, every generator and storage unit would follow the zero profit rule. However, the modeling framework developed in this model provide power systems with a range of NPVs for each



Figure 6.12: Histogram of NPV for technologies built in 2030 with bootstrapping 1,000 possible combination of NPVs within each 10-year investment period.

technology. This is due to the fact that the modeling framework considers a myopic, imperfect power system, intentionally deviating from a perfect competitive market to help account from the imperfections that occur in the real world power market. Storage technologies have the least favorable business case, with negative NPVs for all alternatives. In 2030, no storage technologies are built in the Dutch power system in the optimal case. However, batteries and/or hydrogen storage are built in some of the MGA alternatives. The results indicate that investments in storage technologies are unfavorable in the 2030 time frame in the Dutch power system.

Figure 6.13 visualizes the same data as Figure 6.12 as box-plots. This alternative graphical representation allows for the affect different MGA alternatives have on each technologies respective NPVs to be seen more clearly. The box plots show that the MGA alternatives that maximize a technology result in the amongst the lowest NPVs for the respective technology, particularly for CCGT, OCGT, offshore wind, solar, and hydrogen. This suggests that when more than the optimal amount of a technology is built, the technology cannibalizes its own revenue. For the case of onshore wind, additional flexibility options in the system, such as increased storage or transmission capacity, lead to higher NPVs. The increased flexibility that storage and transmission provide to the energy system help to mitigate the cannibalization effects that VREs face (Prol, Steininger, and Zilberman, 2020). The ability for storage technologies to arbitrage, purchase electricity and charge when electricity prices are low and discharge and sell electricity when price are high help to stabilize the electricity price and therefore damped the cannibalization effects experienced by VREs. Large distributions within an alternative indicates that the weather and demand year has a relatively larger effect on the revenue for the given technology.



Figure 6.13: Boxplot for each run of NPV for technologies built in 2030 with bootstrapping 1,000 possible combination of NPVs within each 10-year investment period.

# Discussion, research limitations & further research

In this thesis, the goal was to investigate if ESOMs could be used to help aid in energy investment decisions. The modeling framework developed in this research and the case study proof of the concept revealed how energy system optimization modeling can be used for investment decisions. The combination of the range of near-optimal solution for the installed capacities for 2030, 2040, and 2050 paired with the results from the investment analysis provide useful insights into the power system and the power market. Investors can use the combination of these results to help support investment decisions as the energy system evolves to achieve emission reduction targets.

Throughout the research, the necessity and importance of appropriate uncertainty analyzes was very apparent. To be able to account for the variability and uncertainty in the power system and provide useful insights for energy system developments, extensive uncertainty analyzes must be performed. To adequately use ESOMs for energy investment decisions requires a full scale uncertainty analysis. Due to the limitations of the study, only the structural uncertainty was considered and modeled. The analysis focuses on Modeling-to-Generate Alternatives and does not consider parametric uncertainty. To perform a complete assessment of the profitability of the future energy system and gain further insights into investment decisions, a full uncertainty analysis should be performed including a robust uncertainty analysis over the parametric uncertainty. Particularly, the demand and weather year used to perform the capacity optimization should be explored. The affects of various years on the profitability of investments is explored by using various historical weather and demand data in the operations optimization portion of the modeling framework. In the operations optimization, the demand and weather data from 2007-2017 is used to help represent the variability of weather and demand data that exists from year to year and to reveal how, based on a power system configuration, the profitability of different generator and storage units changes with various weather and demand data. However, the determination of the capacity in the model is currently performed over one year - the capacity optimization is performed using one, average weather year, the year 2013. In addition, the demand is modeled to be elastic but correlated to and based on reference values from 2013. Therefore, the optimization determines the optimal installed capacity for 2030, 2040, and 2050 assuming that the weather and

demand (scaled to account for expected growth in electricity demand) for each of the future years is equivalent to 2013. However, in reality, the demand and weather data varies significantly from year to year. The model should be further developed to account for the parametric uncertainty in the demand and weather. The need for the incorporation of uncertainty in the demand can be particularly seen in the drastic reduction in power demand that occurred in the spring of 2021 due to Covid-19. The high levels of renewable generated electricity combined with reduced power demand from Covid-19, led to record levels of negative electricity prices (Halbrügge et al., 2021). Models used to forecast the future should be able to account for instances as unforeseen as Covid-19 or for periods of extremely low wind and solar production, such as the Dunkelflaute. In addition, modeling the uncertainty in weather data is becoming increasing important as climate change is leading to more extreme weather conditions across our global, having significant impacts on both the generation of solar and wind generators and on the power demand (particularly the heating and cooling of buildings). Increased storage capacities will be needed to account for extreme weather conditions.

Another essential input parameter to consider the uncertainty of is the technology cost inputs. Singular cost projects are currently considered for all fuel prices and technologies. In reality, there are a large amount of uncertainty in the future fuel and technology prices in the 2030-2050 time frames due to uncertainty in the learning curves of technologies, technology uptake, and the cost of raw materials used to make the generators, storage units, or transmission lines.

In addition, the carbon emission constraint should be explored. The current study is performed under the assumption that government outlined emission reduction targets are achieved. The power system configurations generated all met carbon emission reduction targets. However, in reality this is likely to not exactly be the case. To get a full view of investments in the future power system, the carbon emission constraints should be explored. In addition, although the MGA uncertainty analysis is performed, identifying a range of near optimal solutions that can help account for the development of systems that fall outside of the cost optimal solution, scenarios that cover for redundancy in the system for situations that compromise the security of supply (e.g. irregular weather events, unusual disruptions in supply) or for political/strategic reasons (e.g. energy independence, power reserves) are not explicitly explored. Additional constraints and/or relaxed constraints can be added to the model to help account for these factors (e.g. constraints that ensure a country produces a certain percentage of their consumed electricity can ensure a certain level of energy security). Within the uncertainty analysis performed, MGA, additional slack values that bound the feasible space can be explored to see how the model results differ under different cost deviations from the optimal cost solution. Relaxing the cost slack values leads to a wider range of possible system configurations, accounting for a wider range of factors, outside of cost, that influence the system. Tightening the cost slack value leads to a less diverse set of alternative power systems, and therefore leads to a range of near cost optimal solutions that are closer to the cost optimal solution.

The case study is a very simplified 2 node system to show the methodology developed in practice. To be able to drawn more accurate and deeper insights into the investments for the Dutch grid, a more expansive and detailed model would need to be developed or an already developed model could be used. The modeling of only two countries, the Netherlands and Germany, under-represents the flexibility provided by surrounding countries and underplays the possible flexibility provided by interconnection with other countries. In reality, the Netherlands' interconnection with other surrounding countries (i.e. Norway, Belgium, Denmark, the UK) and the potential for increased interconnection allows for additional flexibility options in the system. The interconnection with the other surrounding countries can play an
important role on the Dutch grid and therefore, the power systems of surrounding countries should be modeled as well, with the most ideal scenario being modeling the entire European power system.

In addition to a spatial resolution of 2 nodes (one node per country), and a temporal resolution of 1 hour over the course of a full year are modeled in the case study. The model uses an hourly temporal resolution to help account for time dependent nature of VREs. The 2-node system was chosen for simplicity and computational limitations. There are significant trade-offs between spatial and temporal resolution and computation time. Energy system optimization modeling is computationally intensive (Hörsch and Calitz, 2017). Increasing the number of nodes (i.e. spatial resolution) helps expose transmission bottlenecks and variation in renewable resources but requires more computation time (Hörsch and Calitz, 2017). Low spatial resolution underestimates total cost by ignoring transmission bottlenecks that either require transmission upgrades or restrict welfare-enhancing transfers and lowers average capacity factors by average renewable resources over a large area (Hörsch and Calitz, 2017). Clustering the spatial resolution to one node per country smooths the variability in VRE production to appear less variable than is the case in the real power system with many nodes. In high spatial resolution models, sites with high renewable resource capacities can be fully exploited (Hörsch and Calitz, 2017). Therefore, to help overcome these limitations, higher spatial and temporal resolution can be used. Further research should be done into reducing necessary computation power needed to solve the problem or ways to use more computation power, such as high distributed optimization methods. This is essential when considering modeling a larger spatial area and performing complete uncertainty analyses.

The model considers 10-year investment periods and new capacities can only be built at the being of each 10-year segment. In reality, investment can occur at any point in time. To improve upon the model and the analysis, further research should look to decrease the length of the investment period to more accurately represent the real power system. In addition, the operations model could be more detailed and advanced. To do so, a key element that should be considered in further research is the inclusion of must-run generators in the model and allowing negative electricity prices. As has been shown, particularly over the past couple of years, the increasing share of VREs and must-run conventional generators have lead to increasingly more instances of negative electricity prices in the Netherlands and Germany. In addition, to more accurately represent the real electricity system, the start-up and shut-down of generators and randomly causing generators to go offline, inline with the probability of unexpected shutdowns in the actual power system should be incorporated. Ancillary services are not considered in this analysis. The incorporation of these service could lead to higher revenues for investors than is shown.

In terms of the modeling the elasticity of demand, there are several data related challenges associated with the demand elasticity factors used in ESOMs. As (Decarolis et al., 2017) identifies that these factors can significantly influence the results and therefore the high uncertainty regarding these factors must be taken into consideration. Considering variations in the demand elasticity factors is outside the scope of this study but should be taken into consideration in future research to determine the effect these factors have on model results. Various historic demand data can be used to help do so. Adjusting these factors would help to make the electricity price more accurately represent real world electricity prices and therefore provide more accurate NPVs forecasts.

Power system optimization models are becoming expanded to incorporate cross-sector integration. The analysis only considers the electricity sector. To enhance the research, the coupling of energy sectors (i.e. electricity, heat, transport) should be examined, particularly due to the added flexibility coupling various sectors can have to help overcome electricity grid balancing problems. A model that

combines both cross-boarder and cross-sector integration would provide a better representation for the market, especially as the transition is made to integrate power and heat and increase the penetration of electricity and hydrogen vehicles. This type of modeling is currently being done with models such as PyPSA-Eur-Se-30, which is cross-sector integrated model that builds on the PyPSA-Eur power sector only model.

The complexity, uncertainty, and rapidly evolving nature of the power system demands continuous research and development. All further research and development topics listed above increase the complexity of the optimization problem and therefore, increase the required computation time and power. The balance between creating a model that encompasses the complexity of the real world to give useful, accurate insights and a model that requires a reasonable amount of computation time and can provide a solution in a reasonable amount of time, needs to be considered in all future development. Smart techniques need to be developed to incorporate the topics listed above while still creating a feasible, usable model.

8

## **Conclusion & recommendations**

Energy system optimization modeling is a widely used bottom-up modeling technique to help in the transition to sustainable energy systems. Open source energy optimization tools (such as PyPSA, OSeMOSYS, TEMOA) have become increasingly more advanced and have been cited to be mature enough to be used for decision making capabilities (Groissböck, 2019). The wide availability of open source energy optimization tools leads to the question of how these tools can be utilized from the perspective of the investor to analyze the profitability of different technologies. Therefore, to answer this guestion and fill the gap, this study explores using optimization modeling from the perspective of an investor in the power system. The modeling framework developed in this study utilizes energy system optimization modeling paired with the uncertainty technique, Modeling-to-Generate Alternatives and a financial model to analyze the profitability of investments in the power system. The modeling framework addresses the need to account for both the long-term investment decisions in the power system and the short-term detailed dispatch decisions. Based on the framework, a model is developed. The model is used as a proof of the method developed and is utilized in a case study to see the practical application of the method for the Dutch power system in the 2030 time frame. The research shows that energy system optimization modeling can aid in investment decisions by providing a range of possible future energy systems and therefore the range of technologies that will be required to achieve emission reduction targets and by performing as an electricity price generator. By producing possible future electricity prices and dispatch of generator and storage units, the models can be used to determine the profitability of different technologies. Applying the developed method to a Netherlands case study provides important insights into investments in the Dutch electricity system.

### Main findings:

**Energy optimization modeling is an important tool for investors in the power system and can be utilized to help aid in investment decision making.** With the increasingly advanced ESOMs available, these modeling tools can be a useful addition to an investors set of tools to aid in the decisionmaking process. The modeling framework developed in this research can help do so.

By providing a range of near optimal energy system solutions, the uncertainty technique, MGA can provide valuable insight into the possible future power systems. Energy system optimization

models determine the cost optimal or maximum society welfare system given a set of input parameters. However, in reality cost optimal systems are not achieved due to the imperfections in the market, market failures, policy failures, and uncertainty. To gain insight into the range of possible future energy systems, the MGA technique can be used to generate a range of near-optimal solutions.

There are a wide range of possible near optimal future power systems. This research reveals that small deviations in overall cost of an energy systems can lead to significantly different energy infrastructure designs. This is crucial to understand for energy modelers, energy investors, and policy makers, as small cost assumptions (deviations) can lead to vastly different energy system designs, business cases and adequate policy to achieve governmental targets. The range of possible energy systems provides flexibility and opportunity for both investors and policy makers. Knowing and examining these possible ranges allow policy makers to incorporate and consider other factors that effect the power system aside from cost.

Near-optimal solutions generated by energy system optimization modeling do not necessarily ensure the economic feasibility of investments required to achieve the respective energy system. The economic feasibility of investments in the range of near optimal electricity systems are not ensured. The economic feasibility of investments in such systems should be studied in parallel with energy system optimization models.

**Main findings of case study:** These findings should be interpreted with the understanding of the limitations of the model developed in the study. The model developed is a simplified model whose main purpose is to explore and provide a proof of concept of the modeling framework developed in this research. The main findings listed below show the findings that can be derived from the utilization of the modeling framework.

Significant investments in the power market are required to meet government set emission reduction targets. A minimum of approximately 4 times the currently installed capacity is required for a carbon neutral power system. Large changes to the power system and substantial capital investments are required to achieve the outlined goals, making the investment perspective on the power system essential to achieving the emission reduction goals.

**Investments in storage technologies in the 2030 time frame are unfavorable.** Although some near-optimal power systems include storage technologies, the business case for such technologies is unfavorable across all scenarios. However, these dynamics could change if governments introduce policy mechanisms to increase the penetration of storage. Governments might implement these type of policies to help drive down storage costs, as they will be needed in the long run to achieve government emission reduction targets.

Storage technologies can help improve the business case for VRE technologies. The presence of storage technologies in the power system in 2030 is shown to potentially increase the NPVs of onshore wind and solar. Storage allows the transfer of load from period of low prices to high prices, flattening the hourly electricity prices and reducing the cannibalization effect experienced by VREs. However, the business case for storage technologies are shown to be poor and therefore, are unlikely to be heavily invested in without governmental support.

#### **Recommendations:**

ESOMs should not be used to give exact results, rather a range of possible near-optimal solutions should be reported. ESOM like all other modeling should be used to give insights, not exact results or even numerical outputs. Giving exact, optimal results like in the case of a basic ESOM, gives a false sense of certainty about the future and the capability of modeling the future. The uncertainty analysis, MGA can be a very useful technique to enrich the results of an ESOM and provide more robust data based support for decision makers.

Policy makers need to consider how favorable investments are to investors, and therefore how likely the investments are to occur, to ensure adequate investments are made to reach emission reduction targets and ensure security of supply. The case study results reveal that the investment environment for VRE and storage technologies in the 2030 time frame in the Netherlands is not particularly appealing. Therefore, to reach government climate targets while ensuring security of supply might require government intervention.

**Caution should be taken to not over-invest in VRE or storage technologies.** The case study analysis reveals that the MGA alternative generated by maximizing a particular technology, leads to the among the lowest NPVs for the respective technology. Therefore, overcapacity of a technology cannibalizes its own revenue.

### **Recommendations for further research:**

**Complexity should be added to the overall model.** The model should incorporate all interconnected countries to capture the impact of transmission. The operation model should be more advance. Must run generators, hydro-power, start-up and shut-down of generators, generators outages, and ancillary services should be incorporated in the operation model.

**Cross-sector integration should be incorporated into the model.** To enhance the research, the coupling of energy sectors (i.e. electricity, heat, transport) should be examined, particularly due to the added flexibility coupling various sectors can have to help overcome electricity grid balancing problems. A model that combines both crossboarder and crosssector integration would provide a better representation for the market, especially as the transition is made to integrate power and heat and increase the penetration of electricity and hydrogen vehicles.

**Demand elasticity should be further explored.** There are several data related challenges associated with the demand elasticity factors used in ESOMs. The demand elasticity factors can significantly influence the results and therefore the high uncertainty regarding these factors should be explored and taken into consideration in future work.

**Parametric uncertainty should be performed on the model.** The current framework only considers structural uncertainty; the MGA analysis changes the underlying structure of the mathematical model to find the near-optimal power system portfolios. To be able to provide robust decision making support, a parametric uncertainty analysis should be performed.

**Methods to reduce computation time should be explored.** In the modeling the energy system, it is essential to find a balance between the required computational power and the accuracy of the model. Given the long time periods and extensive model and parametric uncertainty, energy system optimization modeling quickly encounters computational power and time constraints. Therefore, techniques to reduce computation time should be explored in parallel with the above mentioned recommendations to allow for feasible computation power and time requirements.

# $\bigcirc$

## Reflection

Throughout the course of the research, I often struggled with the idea of how best to utilize an optimization model for a purpose that deviates from its objective function. ESOM's objective are to either maximize social economic welfare or minimize total system costs. The goal of this research was to determine how an investor can utilize ESOMs. An investors objective is to maximize profit which, given the imperfections in the market, do not necessarily correlate to maximizing social economic welfare or total system cost minimization. Under perfect competitive market conditions, producers and consumers behaving in a fashion to maximize their surplus would result in a system that maximizes social welfare and results in a zero net profit for producers. However, due to market imperfections, this is often not the case.

Several times, I ran into difficulties with the difference between theory and reality in the energy system, both conceptually and how to use a modeling technique, that is typically rooted in how the energy system should theoretically operate, to adequately forecast the range of possible realistic futures of the energy system. The uncertainties and complexities of the power system cause modeling the future system to be a very challenging task. As DeCarolis et. al stated:

Given the complexity of the modeled system and the inability to validate model results, energy modeling requires a significant amount of modeler judgment that – depending on one's perspective – makes energy modeling a blend of art and science or a craft that is neither art nor science (Decarolis et al., 2017).

After the months spent trying to understand, develop, and utilize energy system modeling, this quote resonated with me and I believe is an accurate statement, exemplifying the complexity and challenges of energy system modeling and the importance of the modeler and their fundamental understanding of both the energy system and the models used. The necessity of having a very firm and solid fundamental understanding in the energy system and optimization modeling became very apparent to me during the first months of the project. This need led to a longer project timeline than originally plan, but I feel fortunate to have been able to invest the time and effort to learn and explore the power system and optimization modeling. This understanding allowed me to be more confident in the 'modeler judgement'

that I had to use during the course of the research. It should be stated however, that the need for modeler's judgement does not undermine the importance and necessity of energy system modeling and their results, it rather amplifies the need of modelers who are well versed in the topic and in the continue development and advancement of energy system modeling to help as we transition to a sustainable energy system.

On a more personal level, being a novice to python and energy optimization modeling prior to starting this project lead to a very steep learning curve to overcome. In addition, the location and state of the world in which I carried out this thesis differed greatly from what I anticipate. I began this thesis with the intention of performing my research and day to day at the Eneco office. In light of Covid-19 and the associated uncertainty and global lock-down, I decided it was best to be with my family in California. I feel very fortunate to have had the opportunity to be with my direct family and close to my family and friends in California during these tumultuous times, but it also lead to some difficulties being very removed from the Netherlands and being 9-hours behind while carrying out my thesis.

I believe that the individual nature of thesis projects is often a difficult component. The innate difficulties of an individual project, paired with the isolation of Covid-19 certainly led to a testing period of time. The limitation of not being able to meet in person made me greatly appreciative and realize the necessity of in person interaction with colleagues when performing researching projects or general work, but it also led me to understand the necessity of being flexible and adaptive, and I hope strengthened my ability to be adaptive in future circumstances.

The past year has been a very stressful, uncertain, and trying year between a global pandemic and social and political unrest. I believe performing my thesis in this context has pushed me and taught me far more than I could have imagined. I greatly enjoyed expanding my knowledge in the technical components of the project: power systems, energy optimization modeling, electricity markets, and programming. In addition, I believe I have grown to learn how to better structure a research project from start to end, how to most effectively self-teach myself topics or skills I had little or no prior experience with, and to become better at being comfortable with and knowing how and when to ask for help or guidance. I look forward to utilizing these skills in my future work.



# Data

Table A.1: Greenhouse gas emissions from electricity generation for the Netherlands and Germany for 2015, 2020, 2030, 2040,
2050 (United Nations, 2020; Federal Ministry for the Environment and Safety, 2020; CBS, 2019; Umweltbundesamt, 2020).

Year	Greenhouse gas emissions $(MTCO_2)$				
	Germany	Total			
1990	366	39.6	406		
2010	313	52.0	365		
2015	304	53.3	357		
2020	200	29.8	230		
2030	139	14.1	153		
2040	69.5	7.05	76.6		
2050	0	0	0		

 Table A.2: Potential renewable energy capacities for Germany and the Netherlands used in model (Brown, Hörsch, and

 Schlachtberger, 2018)

	Potential Renewable Capacity per Country (MW)				
	Solar	Onshore wind	Offshore wind		
Netherlands	46,300	44,100	151,000		
Germany	360,000	452,000	90,400		

2

1

		Year		
	2030	2040	2050	Source
Lifetime (years)				
Coal	40	40	40	(IEA, 2020)
CCGT	30	30	30	(IEA, 2020)
OCGT	30	30	30	(IEA, 2020)
Nuclear	45	45	45	(Schröder et al., 2013)
Onshore wind	30	30	30	(Technology Data, 2020)
Offshore wind	30	30	30	(Technology Data, 2020)
Solar	25	25	25	(IEA, 2020)
Battery	15	15	15	(Cole and Frazier, 2019)
Battery inverter	15	15	15	(Budischak et al., 2013)
Fuel cell	20	20	20	(Budischak et al., 2013)
Electrolysis	25	27	28	(Smolinka et al., 2018)
Transmission (HVAC overhead)	40	40	40	(Zappa et al., 2019)
Investment (FLIR/kW/el)				
	1400	1400	1400	(Schröder et al. 2013)
	820	820	820	(Schröder et al., 2013)
	410	410	410	
Nuclear	6450	6450	410	(Schröder et al., 2013)
	1040	0450	0450	(Schröder et al., 2013)
	1040	960	960	(Technology Data, 2020)
	1570	1450	1420	(Technology Data, 2020)
Solar	000	510	460	(Schröder et al., 2013)
Battery	200	170	150	(Cole and Frazier, 2019)
Battery inverter	380	310	280	(Cole and Frazier, 2019)
Fuel Cell	340	310	290	(Budischak et al., 2013)
	600	540	490	(Smolinka et al., 2018)
	1000	1000	1000	(Hagspiel et al., 2014)
Fixed operating & maintenance (FOM) (%/year)				
Coal	1.9	1.9	1.9	(Schröder et al., 2013)
CCGT	2.5	2.5	2.5	(Schröder et al., 2013)
OCGT	3.8	3.8	3.8	(Schröder et al., 2013)
Onshore wind	1.2	1.2	1.2	(Technology Data, 2020)
Offshore wind	1.9	1.8	1.8	(Technology Data, 2020)
Solar	2.0	2.0	2.0	(Ioannis Tsiropoulos, Dalius Tarvydas, and Andreas Zucke
Battery inverter	3	3	3	(Cole and Frazier, 2019)
Fuel cell	3	3	3	(Budischak et al., 2013; Steward, 2009)
Electrolysis	3.3	3.6	3.9	(Smolinka et al., 2018)
Transmission (HVAC overhead)	2	2	2	(Hagspiel et al., 2014)
Variable operating & maintenance (VOM) (EUR/MWel)				
Coal	6.0	6.0	6.0	(Schröder et al., 2013)
CCGT	4.0	4.0	4.0	(Schröder et al., 2013)
OCGT	3.0	3.0	3.0	(Schröder et al., 2013)
Nuclear	8.0	8.0	8.0	(Schröder et al., 2013)
Onshore wind	1.4	1.2	1.2	(Technology Data, 2020)
Offshore wind	2.7	2.5	2.4	(Technology Data, 2020)
Solar	0.01	0.01	0.01	(Ioannis Tsiropoulos, Dalius Tarvydas, and Andreas Zucke
Efficiency (0/)				
	0.40	0.47	0.47	
	0.46	0.47	0.47	(Schröder et al., 2013)
	0.0	C.U	0.40	(Schröder et al., 2013)
	0.39	0.40	0.40	
	0.34	0.34	0.34	(Schröder et al., 2013)
	0.01	0.00	0.81	(Budischak et al., 2013)
	0.58	0.62	0.62	(Budischak et al., 2013; Steward, 2009)
Electrolysis	0.65	0.66	0.69	(Smolinka et al., 2018)

		Year		
	2030	2040	2050	Source
Fuel price (EUR/MWh <sub>th</sub> )				
Coal	10.35	10.60	10.85	(International Energy Agency, 2018)
Gas	24.33	26.70	29.08	(International Energy Agency, 2018)
Nuclear fuel (uranium)	3.02	3.02	3.02	(Schröder et al., 2013)
Carbon intensity $(tCO_2/MWh_{th})$				
Coal	0.51	0.51	0.51	(Skone et al., 2016)
Gas	0.31	0.31	0.31	(Skone et al., 2016)
Value of loss load (EUR/MWh)				
VOLL	5,000	5,000	5,000	Brown2018a

Table A.4: Fuel price, carbon intensity, and value of loss load forecasts for 2030, 2040, and 2050 in Europe.

		Year		
	2010	2015	2020	Source
Netherlands (NL)				
Hard coal	2943	7270	4662	(Gotzens et al., 2019)
Lignite	0	0	0	(Gotzens et al., 2019)
CCGT	12271	13582	13582	(Gotzens et al., 2019)
OCGT	3991	3991	3991	(Gotzens et al., 2019)
Nuclear	492	492	492	(Gotzens et al., 2019; ENTSO-E, 2020)
Biomass	1205	400	490	(Rijksoverheid, 2010; ENTSO-E, 2020)
Onshore wind	2009	2646	3973	(Rijksoverheid, 2010; ENTSO-E, 2020)
Offshore wind	228	228	1709	(Rijksoverheid, 2010; ENTSO-E, 2020)
Solar	88	1000	5710	(Rijksoverheid, 2010; ENTSO-E, 2020)
Germany (DE)				
Hard coal	28390	28650	22630	(Gotzens et al., 2019; Fraunhofer ISE, 2020)
Lignite	21340	21420	20860	(Gotzens et al., 2019; Fraunhofer ISE, 2020)
CCGT	18121	18121	17256	(Gotzens et al., 2019)
OCGT	7801	7588	6628	(Gotzens et al., 2019)
Nuclear	20500	10800	8110	(Gotzens et al., 2019; Fraunhofer ISE, 2020)
Biomass	6130	7170	8240	(Fraunhofer ISE, 2020)
Onshore wind	26820	41300	54640	(Fraunhofer ISE, 2020)
Offshore wind	80	3280	7740	(IRENA, 2013; ENTSO-E, 2020; Fraunhofer ISE, 2020)
Solar	18000	39220	53580	(Fraunhofer ISE, 2020)
		-		

Table A.5: Historical installed generator capacities in the Netherlands and Germany for 2010, 2015, and 2020.

4



# Research flow chart





Figure B.1

# $\bigcirc$

# Validation dispatch curves

Below are the price duration curves for each of the various years for each alternative is plotted against the historical values.



Figure C.1: Price duration curves for each 10-years (one year for each operations optimization run with a different demand and weather year) of alternative MGA runs (blue curves) plotted against the historical price duration curve for 2019 (green curve).

# Bibliography

- Abrell, Jan, Patrick Eser, Jared B Garrison, Jonas Savelsberg, and Hannes Weigt. 2019. "Integrating economic and engineering models for future electricity market evaluation: A Swiss case study." *Energy Strategy Reviews* 25:86–106. ISSN: 2211-467X. https://doi.org/https://doi.org/ 10.1016/j.esr.2019.04.003. http://www.sciencedirect.com/science/article/pii/ S2211467X19300306.
- Agency, International Energy. 2007. "Climate policy uncertainty and investment risk." *Climate Policy Uncertainty and Investment Risk* 9789264030:1–142. https://doi.org/10.1787/9789264030 152-en.
- Bahar, Heymi, and Jehan Sauvage. 2013. "Cross-Border Trade in Electricity and the Development of Renewables-Based Electric Power." OECD Trade and Environment Working Papers, nos. 2013/02, https://doi.org/10.1787/5k4869cdwnzr-en. http://dx.doi.org/10.1787/5k4869cdwnz r-en.
- Bazilian, Morgan, Andrew Rice, Juliana Rotich, Mark Howells, Joseph Decarolis, Stuart Macmillan, Cameron Brooks, Florian Bauer, and Michael Liebreich. 2012. "Open source software and crowdsourcing for energy analysis." *Energy Policy* 49:149–153. ISSN: 0301-4215. https://doi.org/ 10.1016/j.enpol.2012.06.032. http://dx.doi.org/10.1016/j.enpol.2012.06.032.
- Berntsen, Philip B., and Evelina Trutnevyte. 2017. "Ensuring diversity of national energy scenarios: Bottom-up energy system model with Modeling to Generate Alternatives." *Energy* 126:886–898. ISSN: 03605442. https://doi.org/10.1016/j.energy.2017.03.043. http://dx.doi.org/10. 1016/j.energy.2017.03.043.
- Botterud, Audun. 2003. "Long-Term Planning in Restructured Power Systems: Dynamic Modelling of Investments in New Power Generation under Uncertainty by." PhD, Norwegian University of Science and Technology (NTNU).
- Brill, E. Downey, Shoou Yuh Chang, and Lewis D. Hopkins. 1982. "Modeling To Generate Alternatives: the Hsj Approach and an Illustration Using a Problem in Land Use Planning." *Management Science* 28 (3): 221–235. ISSN: 00251909. https://doi.org/10.1287/mnsc.28.3.221.
- Brown, T., and L. Reichenberg. 2020. "Decreasing market value of variable renewables is a result of policy, not variability," arXiv: 2002.05209. http://arxiv.org/abs/2002.05209.
- Brown, Tom. 2020. *Energy System Modelling Lectures Summer Semester 2020.* https://nworbmot. org/courses/esm-2020/.
- Brown, Tom, Jonas Hörsch, and David Schlachtberger. 2018. "PyPSA : Python for Power System Analysis." Open Research Software 6 (4). https://doi.org/[1]T.Brown,J.Hörsch,andD. Schlachtberger, "PyPSA: PythonforPowerSystemAnalysis, "OpenRes. Softw.,vol.6,no.4, 2018..

- Budischak, Cory, Deanna Sewell, Heather Thomson, Leon MacH, Dana E. Veron, and Willett Kempton. 2013. "Cost-minimized combinations of wind power, solar power and electrochemical storage, powering the grid up to 99.9% of the time." *Journal of Power Sources* 225:60–74. ISSN: 03787753. https://doi.org/10.1016/j.jpowsour.2012.09.054. http://dx.doi.org/10.1016/j.jpowsour.2012.09.054.
- Capros, Pantelis, Maria Kannavou, Stavroula Evangelopoulou, Apostolos Petropoulos, Pelopidas Siskos, Nikolaos Tasios, Georgios Zazias, and Alessia Devita. 2018. "Outlook of the EU energy system up to 2050 : The case of scenarios prepared for European Commission's " clean energy for all Europeans " package using the PRIMES model." *Energy Strategy Reviews* 22 (March): 255–263. ISSN: 2211-467X. https://doi.org/10.1016/j.esr.2018.06.009. https://doi.org/10.1016/ j.esr.2018.06.009.
- CBS. 2019. "Klimaat en Energieverkenning 2019," 242. www.pbl.nl/kev.
- Child, Michael, Claudia Kemfert, Dmitrii Bogdanov, and Christian Breyer. 2019. "Flexible electricity generation, grid exchange and storage for the transition to a 100 % renewable energy system in Europe." *Renewable Energy* 139:80–101. ISSN: 0960-1481. https://doi.org/10.1016/j.renene.2019.02.077. https://doi.org/10.1016/j.renene.2019.02.077.
- Cole, Wesley, and A Will Frazier. 2019. Cost Projections for Utility- Scale Battery Storage Cost Projections for Utility- Scale Battery Storage. Technical report June. Golden, CO: National Renewable Energy Laboratory. https://www.nrel.gov/docs/fy19osti/73222.pdf.
- Conejo, Antonio J, Luis Baringo, S. Jalal Kazempour, and Afzal S. Siddiqui. 2016. *Investment in Electricity Generation and Transmission: Decision Making under Uncertainty.* Basel: Springer. ISBN: 9783319294995.
- Davis, Paul K., Steven C. Bankes, and Michael Egner. 2007. *Enhancing Strategic Planning with Mas*sive Scenario Generation - Theory and Experiments. ISBN: 9780833040176.
- DeCarolis, J. F., S. Babaee, B. Li, and S. Kanungo. 2016. "Modelling to generate alternatives with an energy system optimization model." *Environmental Modelling and Software* 79:300–310. ISSN: 18736726. https://doi.org/10.1016/j.envsoft.2015.11.019.
- Decarolis, Joseph, Hannah Daly, Paul Dodds, Ilkka Keppo, Francis Li, and Will Mcdowall. 2017. "Formalizing best practice for energy system optimization modelling."
- DeCarolis, Joseph F. 2011. "Using modeling to generate alternatives (MGA) to expand our thinking on energy futures." *Energy Economics* 33 (2): 145–152. ISSN: 01409883. https://doi.org/10. 1016/j.eneco.2010.05.002. http://dx.doi.org/10.1016/j.eneco.2010.05.002.
- Deng, Xu, and Tao Lv. 2020. "Power system planning with increasing variable renewable energy : A review of optimization models." *Journal of Cleaner Production* 246:118962. ISSN: 0959-6526. https://doi.org/10.1016/j.jclepro.2019.118962. https://doi.org/10.1016/j.jclepro. 2019.118962.
- Directorate-General for Economic and Financial Affairs. 2015. Energy economic developments: Investment perspectives in electricity markets. Technical report July. https://doi.org/10.2765/48569. http://ec.europa.eu/economy%7B%5C\_%7Dfinance/publications/eeip/ip003%7B%5C\_%7Den.htm.

- Dreier, Dennis, and Mark Howells. 2019. "Osemosys-pulp: A stochastic modeling framework for longterm energy systems modeling." *Energies* 12 (7). ISSN: 19961073. https://doi.org/10.3390/ en12071382.
- ENTSO-E. 2020. Transparency Platform. https://www.entsoe.eu/.
- Federal Ministry for the Environment, Nature Conservation, and Nuclear Safety. 2020. Roadmap to a climate-neutral Germany Climate Action Plan 2050 Germany's long-term low greenhouse gas emission development strategy. https://www.bmu.de/en/topics/climate-energy/climate/national-climate-policy/greenhouse-gas-neutral-germany-2050/%7B%5C#%7Dc 12744.
- Fleiter, Tobias, Ernst Worrell, and Wolfgang Eichhammer. 2011. "Barriers to energy efficiency in industrial bottom-up energy demand models—A review." *Renewable and Sustainable Energy Reviews* 15 (6): 3099–3111. ISSN: 1364-0321. https://doi.org/https://doi.org/10.1016/j.rser. 2011.03.025. http://www.sciencedirect.com/science/article/pii/S1364032111001286.
- Foley, A M, B P Ó Gallachóir, J Hur, R Baldick, and E J Mckeogh. 2010. "A strategic review of electricity systems models." *Energy* 35:4522–4530. ISSN: 0360-5442. https://doi.org/10.1016/j. energy.2010.03.057. http://dx.doi.org/10.1016/j.energy.2010.03.057.
- Fraunhofer ISE. 2020. Energy Charts: Net installed electricity generation capacity in Germany.
- Fuss, Sabine, Jana Szolgayova, Michael Obersteiner, and Mykola Gusti. 2008. "Investment under market and climate policy uncertainty." *Applied Energy* 85 (8): 708–721. ISSN: 03062619. https: //doi.org/10.1016/j.apenergy.2008.01.005.
- Gioutsos, Dean Marcus, Kornelis Blok, Leonore Van Velzen, and Sjoerd Moorman. 2018. "Cost-optimal electricity systems with increasing renewable energy penetration for islands across the globe." *Applied Energy* 226 (April): 437–449. ISSN: 0306-2619. https://doi.org/10.1016/j.apenergy. 2018.05.108. https://doi.org/10.1016/j.apenergy.2018.05.108.
- Gotzens, Fabian, Heidi Heinrichs, Jonas Hörsch, and Fabian Hofmann. 2019. "Performing energy modelling exercises in a transparent way - The issue of data quality in power plant databases" [in en]. *Energy Strategy Reviews* 23 (January): 1–12. ISSN: 2211467X, accessed December 3, 2018. https://doi.org/10.1016/j.esr.2018.11.004. https://linkinghub.elsevier.com/retrieve/ pii/S2211467X18301056.
- Groissböck, Markus. 2019. "Are open source energy system optimization tools mature enough for serious use ?" *Renewable and Sustainable Energy Reviews* 102 (September 2018): 234–248. ISSN: 1364-0321. https://doi.org/10.1016/j.rser.2018.11.020. https://doi.org/10.1016/j.rser.2018.11.020.
- Gross, Robert, William Blyth, and Philip Heptonstall. 2010. "Risks, revenues and investment in electricity generation: Why policy needs to look beyond costs." *Energy Economics* 32 (4): 796–804. ISSN: 01409883. https://doi.org/10.1016/j.eneco.2009.09.017. http://dx.doi.org/10.1016/j.eneco.2009.09.017.
- Hagspiel, S., C. Jägemann, D. Lindenberger, T. Brown, S. Cherevatskiy, and E. Tröster. 2014. "Costoptimal power system extension under flow-based market coupling." *Energy* 66:654–666. ISSN: 03605442. https://doi.org/10.1016/j.energy.2014.01.025. http://dx.doi.org/10.1016/ j.energy.2014.01.025.

- Halbrügge, Stephanie, Paul Schott, Martin Weibelzahl, Hans Ulrich Buhl, Gilbert Fridgen, and Michael Schöpf. 2021. "How did the German and other European electricity systems react to the COVID-19 pandemic?" *Applied Energy* 285 (August 2020): 116370. ISSN: 03062619. https://doi.org/10. 1016/j.apenergy.2020.116370. https://doi.org/10.1016/j.apenergy.2020.116370.
- Hilpert, S, C Kaldemeyer, U Krien, S Günther, C Wingenbach, and G Plessmann. 2018. "The Open Energy Modelling Framework (oemof) - A new approach to facilitate open science in energy system modelling □." *Energy Strategy Reviews* 22 (May 2017): 16–25. ISSN: 2211-467X. https: //doi.org/10.1016/j.esr.2018.07.001. https://doi.org/10.1016/j.esr.2018.07.001.
- Hirth, Lion. 2013. "The market value of variable renewables. The effect of solar wind power variability on their relative price." *Energy Economics* 38 (2013): 218–236. ISSN: 01409883. https://doi.org/10.1016/j.eneco.2013.02.004. http://dx.doi.org/10.1016/j.eneco.2013.02.004.

——. 2017. "EMMA model documentation The European Electricity Market Model EMMA Model documentation," no. 2016, 1–17. http://neon-energie.de/emma.

- Hörsch, Jonas, and Joanne Calitz. 2017. "PyPSA-ZA: Investment and operation co-optimization of integrating wind and solar in South Africa at high spatial and temporal detail," 1–15. arXiv: arXiv: 1710.11199v1.
- Hörsch, Jonas, Fabian Hofmann, David Schlachtberger, and Tom Brown. 2018. "PyPSA-Eur: An Open Optimisation Model of the European Transmission System." *Energy Strategy Reviews*, https: //doi.org/10.1016/j.esr.2018.08.012. arXiv: arXiv:1806.01613v3. https://github.com/ FRESNA/powerplantmatching.
- IEA. 2016. "Re-powering Markets: Market Design and Regulation during the transition to lowcarbon power systems." *International Energy Agency Electricity Market Series*, 246. https://doi.org/ 10.1787/9789264209596-en. https://www.iea.org/publications/freepublications/ publication/REPOWERINGMARKETS.pdf.

——. 2020. "World Energy Model – Analysis," https://www.iea.org/reports/world-energymodel.

- Ifzal, Ali. 1991. *Economic Analysis of Investment in Power Systems*. Technical report. Economic and Development Resource Center, Asian Development Bank.
- International Energy Agency. 2018. "World Energy Outlook 2018." 32 (0): 23–28. https://doi.org/ 10.1787/weo-2018-2-en. https://www.oecd-ilibrary.org/energy/world-energy-outlook-2018/executive-summary%7B%5C\_%7Dweo-2018-2-en.
- Ioannis Tsiropoulos, Dalius Tarvydas, and Andreas Zucker. 2018. Cost development of low carbon energy technologies - Scenario-based cost trajectories to 2050, 2017 edition. ISBN: 9789279774799. https://doi.org/10.2760/23266.
- IRENA. 2013. "30 Years of Policies for Wind Energy: GERMANY." 30 Years of Policies for Wind Energy - Lessons from 12 Wind Energy Markets, 68–75.
- Johnston, Josiah, Rodrigo Henriquez-auba, Benjamín Maluenda, and Matthias Fripp. 2019. "SoftwareX Switch 2.0: A modern platform for planning high-renewable power systems." *SoftwareX* 10:100251. ISSN: 2352-7110. https://doi.org/10.1016/j.softx.2019.100251. https://doi.org/10. 1016/j.softx.2019.100251.

- Kaut, Michal, Kjetil T Midthun, Adrian S Werner, Lars Hellemo, and Marte Fodstad. 2013. "Multi-horizon stochastic programming." *Computational Management Science* 11 (1-2): 179–193.
- Kraan, O., G. J. Kramer, and I. Nikolic. 2018. "Investment in the future electricity system An agentbased modelling approach." *Energy* 151:569–580. ISSN: 03605442. https://doi.org/10.1016/ j.energy.2018.03.092. https://doi.org/10.1016/j.energy.2018.03.092.
- Lempert, Robert J., David G. Groves, Steven W. Popper, and Steve C. Bankes. 2006. "A general, analytic method for generating robust strategies and narrative scenarios." *Management Science* 52 (4): 514–528. ISSN: 00251909. https://doi.org/10.1287/mnsc.1050.0472.
- Loulou, Richard, Gary Goldstein, and Ken Noble. 2004. *Documentation for the MARKAL Family of Models.* Technical report October. International Energy Agency.
- Milligan, Michael, Bethany Frew, Kara Clark, and Aaron Bloom. 2017. "Marginal Cost Pricing in a World without Perfect Competition: Implications for Electricity Markets with High Shares of Low Marginal Cost Resources," no. December.
- Morales, Juan M., Antonio J. Conejo, Henrik Madsen, Pierre Pinson, and Marco Zugno. 2014. *Integrating Renewables in Electricity Markets: Operational Problems.* Edited by Frederick S. Hillier and Camille C. Price. New York: Springer. ISBN: 9781461494102.
- Netherlands Ministry of Economic Affairs (EZK). 2019. *National Climate Agreement of the Netherlands.* Technical report June. The Hague: Ministry of Economic Affairs and Climate Policy. https://www.klimaatakkoord.nl/documenten/publicaties/2019/06/28/klimaatakkoord.
- Neumann, Fabian, and Tom Brown. 2019. "The Near-Optimal Feasible Space of a Renewable Power System Model." *21st Power Systems Computation Conference (PSCC 2020),* no. October, arXiv: arXiv:1910.01891v1.
- Newbery, David, Michael G Pollitt, Robert A Ritz, and Wadim Strielkowski. 2018. "Market design for a high-renewables European electricity system." *Renewable and Sustainable Energy Reviews* 91:695–707. ISSN: 1364-0321. https://doi.org/10.1016/j.rser.2018.04.025. https: //doi.org/10.1016/j.rser.2018.04.025.
- Orgaz, Alberto, Antonio Bello, and Javier Reneses. 2018. "A Monte Carlo approach to represent uncertainty in the European electricity market." *International Conference on the European Energy Market, EEM* 2018-June:1–6. ISSN: 21654093. https://doi.org/10.1109/EEM.2018.8469938.
- Papadelis, S., Alexandros Flamos, and J. Psarras. 2013. "A framework to address uncertainties in energy policy formulation." *Energy Sources, Part B: Economics, Planning and Policy* 8 (3): 270– 278. ISSN: 15567249. https://doi.org/10.1080/15567240903419377.
- Pereira, Sérgio, Paula Ferreira, and A I F Vaz. 2016. "Optimization modeling to support renewables integration in power systems." *Renewable and Sustainable Energy Reviews* 55:316–325. ISSN: 1364-0321. https://doi.org/10.1016/j.rser.2015.10.116. http://dx.doi.org/10.1016/j. rser.2015.10.116.
- Pérez-Arriaga, Ignacio J. 2013. *Regulation of the power sector*, vol. 61. ISBN: 9781447150336. https: //doi.org/10.1007/978-1-4471-5034-3.

- Petitet, Marie. 2017. "Long-term dynamics of investment decisions in electricity markets with variable renewables development and adequacy objectives de l'Université de recherche Paris Sciences et Lettres Préparée à l'Université Paris -Dauphine Long-term dynamics of investm."
- Petitet, Marie, Dominique Finon, and Tanguy Janssen. 2017. "Capacity adequacy in power markets facing energy transition: A comparison of scarcity pricing and capacity mechanism." *Energy Policy* 103 (January): 30–46. ISSN: 03014215. https://doi.org/10.1016/j.enpol.2016.12.032. http://dx.doi.org/10.1016/j.enpol.2016.12.032.
- Pfenninger, Stefan, and Iain Staffell. 2016. "Long-term patterns of European PV output using 30 years of validated hourly reanalysis and satellite data." *Energy*, 1251–1265. https://doi.org/10.1016/j.energy.2016.08.060.
- Poncelet, Kris, Erik Delarue, Daan Six, and William D'Haeseleer. 2016. "Myopic optimization models for simulation of investment decisions in the electric power sector." *International Conference on the European Energy Market, EEM* 2016-July:1–9. ISSN: 21654093. https://doi.org/10.1109/ EEM.2016.7521261.
- Price, James, and Ilkka Keppo. 2017. "Modelling to generate alternatives: A technique to explore uncertainty in energy-environment-economy models." *Applied Energy* 195:356–369. ISSN: 03062619. https://doi.org/10.1016/j.apenergy.2017.03.065. http://dx.doi.org/10.1016/j. apenergy.2017.03.065.
- Prina, Matteo Giacomo, Giampaolo Manzolini, David Moser, Benedetto Nastasi, and Wolfram Sparber. 2020. "Classification and challenges of bottom-up energy system models - A review." *Renewable and Sustainable Energy Reviews* 129:109917. ISSN: 18790690. https://doi.org/10.1016/j. rser.2020.109917. https://doi.org/10.1016/j.rser.2020.109917.
- Prol, Javier López, Karl W Steininger, and David Zilberman. 2020. "The cannibalization effect of wind and solar in the California wholesale electricity market." *Energy Economics* 85:104552. ISSN: 01409883. https://doi.org/10.1016/j.eneco.2019.104552.
- Rijksoverheid. 2010. *Progress report: energy from renewable sources in the Netherlands 2009-2010.* Technical report.
- Ringkjøb, Hans-kristian, Peter M Haugan, and Ida Marie Solbrekke. 2018. "A review of modelling tools for energy and electricity systems with large shares of variable renewables." *Renewable and Sustainable Energy Reviews* 96 (August): 440–459. ISSN: 1364-0321. https://doi.org/10.1016/ j.rser.2018.08.002. https://doi.org/10.1016/j.rser.2018.08.002.
- Schröder, Andreas, Kunz Friedrich, Jan Meiss, Roman Mendelevitch, and Christian von Hirschhausen. 2013. Current and Prospective Costs of Electricity Generation until 2050 2013. Technical report. Deutsches Institut für Wirtschaftsforschung (DIW). https://www.diw.de/documents/publik ationen/73/diw%7B%5C\_%7D01.c.424566.de/diw%7B%5C\_%7Ddatadoc%7B%5C\_ %7D2013-068.pdf.
- Shahnazari, Mahdi, Bryan Maybee, Jonathan Whale, and Adam Mchugh. 2015. "Climate policy uncertainty and power generation investments : A real options-CVaR portfolio optimization approach." In *The 7th International Conference on Applied Energy – ICAE2015 Climate*, 75:2649–2657. https: //doi.org/10.1016/j.egypro.2015.07.367.

- Skone, Timothy J, Strategic Energy Analysis, Planning Division, and National Energy. 2016. Two questions – from a life cycle GHG perspective. Technical report. U.S. Department of Energy: National Energy Technology Laboratory.
- Smolinka, Tom, Nikolai Wiebe, Philip Sterchele, Andreas Palzer, Franz Lehner, Malte Jansen, Kimmelm Steffen, Robert Miehe, Sylvia Wahren, and Fabian Zimmerman. 2018. Study IndWEDe Brief Overview: Industrialisation of water electrolysis in Germany: Opportunities and challenges for sustainable hydrogen for transport, electricity and heat. Technical report. Berlin: Federal Ministry of Transport and Digital Infrastructure (BMVI). https://www.now-gmbh.de/content/service/3publikationen/1-nip-wasserstoff-und-brennstoffzellentechnologie/181204%7B%5C\_ %7Dbro%7B%5C\_%7Da4%7B%5C\_%7Dindwede-studie%7B%5C\_%7Dkurzfassung% 7B%5C\_%7Den%7B%5C\_%7Dv03.pdf.
- Staffell, Iain, and Stefan Pfenninger. 2016. "Using Bias-Corrected Reanalysis to Simulate Current and Future Wind Power Output." *Energy*, no. 114, 1224–1239. https://doi.org/10.1016/j.energy. 2016.08.068.
- Steward, Darlene M. 2009. "Scenario Development and Analysis of Hydrogen as a Large-Scale Energy Storage Medium," https://www.nrel.gov/docs/fy09osti/45873.pdf.
- Tash, Ali, Mohammad Ahanchian, and Ulrich Fahl. 2019. "Improved representation of investment decisions in the German energy supply sector: An optimization approach using the TIMES model." *Energy Strategy Reviews* 26:100421. ISSN: 2211467X. https://doi.org/10.1016/j.esr.2019. 100421. https://doi.org/10.1016/j.esr.2019.100421.
- *Technology Data.* 2020. Danish Energy Agency (DEA). https://ens.dk/en/our-services/projecti ons-and-models/technology-data.
- Thacker, Ben H., Scott W. Doebling, Francois M. Hemez, Mark C. Anderson, Jason E. Pepin, and Edward a. Rodriguez. 2004. "Concepts of Model Verification and Validation." *Concepts of Model Verification and Validation*, 41.
- Trutnevyte, Evelina. 2016. "Does cost optimization approximate the real-world energy transition ?" *Energy* 106:182–193. ISSN: 0360-5442. https://doi.org/10.1016/j.energy.2016.03.038. http://dx.doi.org/10.1016/j.energy.2016.03.038.
- Umweltbundesamt. 2020. Entwicklung der spezifischen Kohlendioxid-Emissionen des deutschen Strommix 1990-2018. https://www.umweltbundesamt.de/bild/entwicklung-der-spezifischenkohlendioxid-1.
- United Nations. 2020. *Greenhouse Gas Inventory Data Detailed data by Partyitle*. https://di.unfccc. int/detailed%7B%5C\_%7Ddata%7B%5C\_%7Dby%7B%5C\_%7Dparty.
- Ventosa, Mariano, Pedro Linares, and Ignacio J. Pérez-Arriaga. 2013. "Power System Economics." Chap. Power Syst in *Regulation of the Power System*, 1st ed., edited by Ignacio J. Pérez-Arriaga. London: Springer. ISBN: 9781447150336. https://doi.org/10.1007/978-1-4471-5034-3.
- Voll, Philip, Mark Jennings, Maike Hennen, Nilay Shah, and André Bardow. 2015. "The optimum is not enough: A near-optimal solution paradigm for energy systems synthesis." *Energy* 82:446–456. ISSN: 03605442. https://doi.org/10.1016/j.energy.2015.01.055.

- Walker, Warren E, Robert J Lempert, and Jan H Kwakkel. 2013. "Deep Uncertainty: Uncertainty in Model-Based Decision Support." *Encyclopedia of operations research and management science* 1 (1): 395–402. https://pdfs.semanticscholar.org/8e6b/c8cd6c880e54c68e6c1c71a9f9a 5a5781283.pdf.
- Winkler, Jenny, Alberto Gaio, Benjamin Pfluger, and Mario Ragwitz. 2016. "Impact of renewables on electricity markets - Do support schemes matter?" *Energy Policy* 93:157–167. ISSN: 03014215. https://doi.org/10.1016/j.enpol.2016.02.049.
- World Energy Council. 2014. 2014 World Energy Issues Monitor: What keeps energy leaders awake at night? Technical report. London: World Energy Council.
- Yue, Xiufeng, Steve Pye, Joseph Decarolis, Francis G N Li, Fionn Rogan, and Brian Ó Gallachóir. 2018. "A review of approaches to uncertainty assessment in energy system optimization models." *Energy Strategy Reviews* 21:204–217. ISSN: 2211-467X. https://doi.org/10.1016/j.esr. 2018.06.003. https://doi.org/10.1016/j.esr.2018.06.003.
- Zappa, William, Martin Junginger, Machteld Van Den Broek, and Corine Land Cover. 2019. "Is a 100 % renewable European power system feasible by 2050 ?" *Applied Energy* 233-234 (January 2018): 1027–1050. ISSN: 0306-2619. https://doi.org/10.1016/j.apenergy.2018.08.109. https://doi.org/10.1016/j.apenergy.2018.08.109.